Power Allocation in a QoS-Aware Cellular-based Vehicular Communication System

by

Sello Leonard Mankge

Submitted in fulfilment of the academic requirements of

Master of Science

in Electronic Engineering

School of Engineering College of Agriculture, Engineering and Science University of KwaZulu-Natal Durban South Africa

March 2021

PREFACE

The research contained in this dissertation was completed by Sello Leonard Mankge while based in the Discipline Electronic Engineering of School of Engineering of the College of Agriculture, Engineering and Science, University of KwaZulu-Natal, Durban, South Africa.

The contents of this work have not been submitted in any form to another university and, except where the work of others is acknowledged in the text, the results reported are due to investigations by the candidate.

Signed: Dr. Farzad Ghayoor Date: 21-03-21

DECLARATION 1: PLAGIARISM

I, Sello Leonard Mankge, declare that:

(i) the research reported in this dissertation, except where otherwise indicated or acknowledged, is my original work;

(ii) this dissertation has not been submitted in full or in part for any degree or examination to any other university;

(iii) this dissertation does not contain other persons' data, pictures, graphs, or other information unless specifically acknowledged as being sourced from other persons;

(iv) this dissertation does not contain other persons' writing unless specifically acknowledged as being sourced from other researchers. Where other written sources have been quoted, then:

a) their words have been re-written but the general information attributed to them has been referenced;

b) where their exact words have been used, their writing has been placed inside quotation marks, and referenced;

(v) where I have used material for which publications followed, I have indicated in detail my role in the work;

(vi) this dissertation is primarily a collection of material, prepared by myself, published as journal articles, or presented as a poster and oral presentations at conferences. In some cases, additional material has been included;

(vii) this dissertation does not contain text, graphics or tables copied and pasted from the Internet, unless specifically acknowledged, and the source being detailed in the dissertation and the References sections.

Signed: Sello Leonard Mankge

Date: 21-03-2021

DECLARATION 2: PUBLICATIONS

DETAILS OF CONTRIBUTION TO PUBLICATIONS that form part and/or include research presented in this dissertation:

Publication 1:

Mankge, S.L., and F. Ghayoor. *Power Allocation for QoS-Aware Underlay Device-to-Device Communications System.* in *Proceedings of the Southern Africa Telecommunication Networks and Applications Conference (SATNAC 2019),* September 2019, pp. 80-85.

Publication 2:

Mankge, S.L., and F. Ghayoor. *Power Allocation for D2D-Enabled Vehicular Communications to Support Driver Assistance Systems*. in *Proceedings of the Institute of Electrical and Electronics Engineers (IEEE) AFRICON*, 2019, September 2019.

The papers were authored by Sello Leonard Mankge and co-authored by Farzad Ghayoor.

Signed: Sello Leonard Mankge

Date: 21-03-2021

ABSTRACT

The task of a driver assistance system is to monitor the surrounding environment of a vehicle and provide an appropriate response in the case of detecting any hazardous condition. Such operation requires real-time processing of a large amount of information, which is gathered by a variety of sensors. Vehicular communication in future vehicles can pave the way for designing highly efficient and cost-effective driver assistance systems based on collaborative and remote processing solutions. The main transmission links of vehicular communication systems are vehicle-to-vehicle (V2V) and vehicle-to-infrastructure (V2I). In this research, a cellular-based vehicular communication system is proposed where Device-to-device (D2D) communication links are considered for establishing V2V links, and cellular communication links are employed for V2I links.

D2D communication is one of the enablers of the next generation of cellular networks for improving spectrum and power utilization. D2D communication allows direct communication between user equipments within a cellular system. Nevertheless, implementing D2D communication should not defect nearby ongoing communication services. As a result, interference management is a significant aspect of designing D2D communication systems. Communication links in a cellular network are supposed to support a required level of data rates. The capacity of a communication channel is directly proportional to the energy of a transmitted signal, and in fact, achieving the desired level of Quality of Service (QoS) requires careful control of transmission power for all the radio sources within a system. Among different methods that are recommended for D2D communications, in-band D2D can offer better control over power transmission sources.

In an underlay in-band D2D communication system, D2D user equipments (DUEs) usually reuse the cellular uplink (UL) spectrum. In such a system, the level of interference can effectively be managed by controlling the level of power that is transmitted by user equipments. To effectively perform the interference management, knowledge of the channel state information is required. However, as a result of the distributed nature of DUEs, such information is not fully attainable in a practical D2D system. Therefore, statistical methods are employed to find boundaries on the allocated transmission powers for achieving sufficient spectral efficiencies in V2I and V2V links without considering any prior knowledge on vehicles' locations or the channel state information. Furthermore, the concepts of massive multiple-input multiple-output and underlay D2D communication sharing the uplink spectrum of a cellular system are used to minimize the interference effect.

ACKNOWLEDGMENTS

Throughout the writing of this dissertation, I have been afforded a great deal of assistance and support. I'm deeply thankful to my supervisor Dr. Farzad Ghayoor for what he has taught me, his invaluable advice, and guidance in helping me grow as a researcher.

To my family, words cannot express my sincere appreciation and gratitude for your support and encouragement through all the challenges that I have experienced in life. Thank you so much, to the all-knowing God, thanks for all the opportunities provided in life.

TABLE OF CONTENTS

Page

PREFACEi
DECLARATION 1: PLAGIARISMii
DECLARATION 2: PUBLICATIONSiii
ABSTRACTiv
ACKNOWLEDGMENTSvi
TABLE OF CONTENTS
LIST OF TABLES
LIST OF FIGURESxi
LIST OF ABBREVIATIONS AND DESCRIPTIONS
Chapter 1 : INTRODUCTION
1.1. Background1
1.2. Problem Statement
1.3. Aims and Objectives
1.4. Outline of Dissertation Structure
Chapter 2 : LITERATURE REVIEW
2.1. Introduction
2.2. D2D Communications Challenges

2.3. Interference Management in D2D Communication Systems
2.3.1. Interference Management in Underlaid D2D Communication Systems 11
2.4. Application of D2D Communications in Vehicular Communications15
2.4.1. Interference Management in D2D-based Vehicular Communications 16
2.5. D2D Communications Performance Evaluation
2.6. Summary
Chapter 3 : MATHEMATICAL MODELS
3.1. Introduction
3.2. Cellular Systems
3.3. The Model of a Wireless Channel
3.3.1. Path Loss and Shadowing23
3.3.2. Small-Scale Fading Model
3.3.3. Noise
3.4. Massive MIMO
3.5. Summary
Chapter 4 : POWER ALLOCATION FOR A QOS-AWARE UNDERLAY D2D
COMMUNICATION SYSTEM
4.1. Introduction
4.2. System Model
4.3. Power Allocation
4.4. Summary

Chapter 5 : POWER ALLOCATION FOR A D2D-ENABLED VEHICULAR
COMMUNICATION SYSTEM
5.1. Introduction
5.2. System Model
5.3. Power Allocation
5.4. Summary
Chapter 6 : RESULTS AND DISCUSSION
6.1. Introduction
6.2. Simulation Results and Discussions
Chapter 7 : CONCLUSION
REFERENCES

LIST OF TABLES

Table	<u>Page</u>
Table 6.1. Simulation parameters	

LIST OF FIGURES

LIST OF ABBREVIATIONS AND DESCRIPTIONS

ABBREVIATION	DESCRIPTION
3GPP	Third Generation Partnership Project
ADAS	Advanced Driver Assistance Systems
BS	Base Station
CSI	Channel State Information
CUE	Cellular User Equipment
D2D	Device-to-Device
DL	DownLink
DUE	Device-to-Device User Equipment
ISM	Industrial Scientific Medical
LoS	Line-of-Sight
LTE	Long-Term Evolution
MIMO	Multiple-Input Multiple-Output
ProSe	Proximity Services
QoS	Quality-of-Service
UE	User Equipment
UL	UpLink
V2I	Vehicle-to-Infrastructure
V2V	Vehicle-to-Vehicle
V2X	Vehicle-to-Everything
VANET	Vehicular Ad hoc NETwork
WAVE	Wireless Access for Vehicular Environments

Chapter 1 : INTRODUCTION

1.1. Background

The desire to build autonomous vehicles and the need to increase the level of road safety has attracted much attention to the research and development of advanced driver assistance systems (ADAS). Such attention has also been paid to other related technologies, such as vehicular communications. A driver assistance system uses a variety of sensors for monitoring a vehicle's surroundings to detect hazardous conditions and so to act accordingly. The response of a driver assistance system, depending on the level of its complexity, can either be passive, such as giving a warning to the driver, or can be active by getting involved in controlling the vehicle. Although nowadays, most high-end vehicles are equipped with a form of a driver assistance system, deployment of such systems on a wide range of vehicles requires a thorough consideration of other factors such as the system's size, weight, and cost [1].

Vehicular communications and remote processing solutions can be considered as enablers for designing cost-effective ADAS in future vehicles. A cost-effective driver assistance system can operate based on a collaboration between nearby vehicles and the edge of a network. As a result, a collaborative driver assistance system requires an effective vehicular communication system to support sharing information among different entities. However, the overall system's performance would be highly dependent on the reliability of communication links.

Cellular-based vehicle-to-everything (V2X) communication links have been proposed by the Third Generation Partnership Project (3GPP) in Release 14 of the enhanced Long-Term Evolution (LTE) radio interface, which is known as LTE-Advanced (LTE-A). A collaborative driver assistance system, which operates based on LTE V2X links, can benefit from the already available and almost ubiquitous coverage of an LTE cellular network [2]. However, the requirements of V2X links, in the sense of data rate and required Quality-of-Service (QoS), for supporting different ADAS applications are yet needed to be identified.

The next generation of cellular networks, which is known as 5G, should support a higher data rate with a better QoS. Therefore, new techniques for improving cellular network capacity are required. Device-to-device (D2D) communications are one of the solutions for improving spectrum and power utilization in the next generation of cellular networks. The 3GPP proposed D2D in the Release 12 of LTE-A as Proximity Services (ProSe). In a D2D communication system, nearby mobile devices can communicate directly without their information traversing the cellular Base Station (BS) or the evolved NodeB (the radio part of an E-UMTS radio transmission site). As the distances between D2D user equipments (DUEs) are relatively short, D2D links can provide low-latency, high data-rate communications while consuming relatively less amount of energy.

Although ProSe is a public safety network feature, the specifications of D2D communications make it a suitable candidate for being used in a cellular-based vehicular communication system. The main links in vehicular communication systems are vehicle-to-vehicle (V2V) and vehicle-to-infrastructure (V2I), which can respectively be supported by D2D and Device-to-eNodeB links in a cellular-based V2X.

1.2. Problem Statement

In this research, a cellular-based vehicular communication system is considered. Vehicles in this system can be part of a V2V or V2I link without any limitation. The communication links between vehicles (V2V links) are formed based on underlay D2D communications. Recently, there has been much research on D2D-based vehicular communication systems to achieve the required QoS in the communication links. The majority of the efforts were on minimizing the effect of D2D communication on the nearby ongoing communication services by considering careful resource management.

Resource management includes controlling the utilized spectrum and the transmission powers to control and mitigate the interference effects. On the spectrum management part, the interference between cellular user equipments (CUEs) and DUEs can be cancelled by separating D2D and cellular communications spectra. However, in such systems, the allocated spectrum for D2D communications can be affected by other interference sources, which are not under the control of the cellular system. Alternatively, when the cellular spectrum is shared between CUEs and DUEs, careful power allocation can be implemented to reduce the effect of interferences.

In this research, we will take an alternative approach for controlling the interference, which is by employing the concept of massive multiple-input multiple-output (MIMO) to minimise the interference effects of user-to-cellular links. This concept has not been considered in D2D-based vehicular communication systems. We then calculate the boundaries on the transmitted power in such a system to obtain the required throughputs for V2V and V2I links. The provided analysis are derived based on statistical averages. Therefore, prior knowledge of vehicle locations or the channel state information (CSI) is not required during calculations as the average of all possibilities is considered as the average performance. In summary, the problem is not to optimize power allocation to maximize the spectral of energy efficiency. Instead, it is to find the power boundaries to achieve the required QoS in communication links.

1.3. Aims and Objectives

The main aim of the research is to derive equations for calculating the minimum transmission power that provides the required spectral efficiency for both CUEs and DUEs, without considering any limitation on the spectrum sharing among devices. As the preference of D2D is for short-distance communications, we only consider D2D links between users that are in each other's line-of-sight (LoS). The calculated boundaries have then been applied to a cellularbased vehicular communication system for calculating the transmission powers to achieve sufficient spectral efficiencies for V2I and V2V links.

The research objectives for this research are:

- Calculating the minimum transmission power that provides the required spectral efficiency for both CUEs and DUEs in a cellular network.
- Calculating boundaries on the V2I and V2V transmission links in a cellular-based vehicular communication system by considering the required spectral efficiencies.

1.4. Outline of Dissertation Structure

Chapter 2 Discusses an in-depth literature review with a focus on D2D communication systems, their advantages, and challenges. We then review the proposed methods for interference management in D2D communication systems. This follows by a review of the existing literature on D2D-based vehicular communication systems and interference management control techniques in such systems.

Chapter 3 Focuses on the mathematical concepts considered for wireless multipath propagation. We then present the concepts of Massive MIMO that will be the base of our analysis in future chapters.

Power allocation for a QoS-aware underlay D2D communication system is discussed in Chapter 4. We used the mathematical concepts that are covered in Chapter 3 to model the system and identify the power boundaries for obtaining the required channel throughputs.

Chapter 5 expands the model given in Chapter 4 to discuss the Power allocation for a D2Denabled vehicular communication system. The main difference is the distribution of the users (vehicles) in such a system compared to the system covered in Chapter 4.

Chapter 6 focuses on the results and discussion. Numerical analyses have been used to demonstrate the effect of transmission power on the V2I and V2V spectral efficiencies.

Chapter 7 presents the conclusion drawn for this study.

Chapter 2 : LITERATURE REVIEW

2.1. Introduction

The demand for accessing information anytime and anywhere and the need for having a low delay, high data rate transmission links have led to a faster rate of advances in the development of cellular technologies. Nowadays, it is common for a user to have multiple devices simultaneously connected to the Internet through cellular and wireless networks. Also, the cellular networks' data-traffic is continuously on the rise, which creates pressure on the existing network resources. The fifth generation of cellular networks should support a communication system with data rates of 1 Gbps and, compared to its previous counterparts, should consume less power and offer better spectral and power efficiencies [3]. Moreover, the 5G should be capable of increasing the storage capacity and computational power of its connected devices by providing remote storage and distant computing capabilities [4]. As a result, employing new techniques for increasing energy and spectral efficiencies is required.

One solution for increasing the energy and spectral efficiencies of a cellular network is through densification, which is done by adding more network resources for reducing the size of the cells [5]. The employment of microcells, picocells, and femtocells in a cellular network are examples of such practice. Although densification will also decrease transmission delays, implementing smaller size cells requires additional infrastructure, which increases the cellular network's cost [6].

D2D communication is an alternative solution for improving spectrum and energy utilization in the next generation of cellular networks [7]. In D2D communication, nearby mobile devices, which are known as DUEs, communicate with one another without their data passing through a cellular BS. As the distances between DUEs are shorter than those between a cellular BS and its users, D2D communications can also reduce overall energy consumption in a cellular network [8] and enables D2D links to use less power for transmitting higher data-rates [9]. Another vital advantage of D2D communication is its capability to enhance coverage of a cellular network on the edges of a cell or in the areas where coverage is weak [10]. A BS can access out-of-coverage equipments through users that receive an adequate level of signal power.

All the specifications mentioned above make D2D communication an essential player in the 5G cellular network and also, different applications can be considered based on the D2D communications concept. Examples of such applications are in sharing large volume contents such as videos and pictures [11], disaster relief [12], gaming [13], and vehicular communications [14]. The D2D communications is also a favourable technology in designing cost-effective ADAS applications for future vehicles [15].

2.2. D2D Communications Challenges

Despite D2D communications' numerous advantages, many issues such as discovery methods, mode selection, resource management, mobility, and security are still needed to be addressed before D2D communications can be fully implemented. The first requirement to establish a D2D communication network is that a device should discover other potential candidates for setting up a direct D2D connection. For this purpose, DUEs should share beacon signals containing information such as a device's ID, location, and channel state information [16]. The potential D2D receivers can then use this acquired information to evaluate the possibility of pairing with one another. The discovery procedure can be done with the help of a central entity, such as a BS (centralized discovery) [17], or can be done among user equipments themselves (distributed discovery) [18].

The next challenge in D2D communications is known as mode selection, which identifies the way that DUEs can communicate with each other. In summary, mode selection deals with the allocation of the frequency spectrum for D2D communication links in a cellular network [19] and also identifies the number of DUEs, which can be involved in establishing a D2D communication network [20]. Pure cellular, partial cellular, underlay/overlay D2D, and single/multiple hop(s) are examples of modes that can be selected for a D2D communication system. The concept of D2D communication is nothing new, and it has existed in different forms of technologies since the earlier generations of mobile devices. Direct Wi-Fi and Bluetooth connections are among these technologies that are implemented over the unlicensed spectrum. D2D communication, as part of a licensed spectrum, was first proposed in [21] to

support multi-hop relays. Other studies showed that D2D communications have the potential to improve, both, spectral and energy efficiency of a cellular network [22]. Nevertheless, it was only in 2012 that the 3GPP proposed D2D communications in the Release 12 of LTE-A as ProSe, which is a public safety network feature [23].

D2D communications, as is shown in Figure 2.1, is categorized based on its operational frequency spectrum into in-band and out-band D2D systems. In the in-band D2D, the communication between D2D devices occurs at the same spectrum as the cellular communication, and in the case of out-band D2D, an unlicensed spectrum is considered for the D2D links.



Figure 2.1. D2D communications classification [24]

The in-band D2D communication is classified as underlay and overlay [25]. In the in-band underlay scheme, DUEs competitively use the complete spectrum allocated to cellular communications. This means the D2D communication links reuse the dedicated resource blocks assigned to the CUEs in a direct communication mode. The in-band underlay D2D communication can improve the spectral efficiency of a cellular network [26]. In the overlay D2D communication system, only a portion of the cellular spectrum is allocated for D2D communication [27]. The portion that is allocated to the D2D communication cannot be used by cellular links. Therefore, the effect of cross-interference will be reduced.

The out-band D2D communication uses the licensed cellular spectrum for CUEs communications and an unlicensed spectrum, which are mostly Industrial Scientific Medical (ISM) bands, for D2D links [7]. Bluetooth and Wi-Fi are examples of technologies, which operate over the ISM band. In the previous generations of cellular networks, the users of such systems have to set up each D2D link manually. However, a better user experience can be obtained if a BS and devices can automatically handle D2D parings. Nevertheless, the main challenge of the out-band D2D is the need to encode and decode data packets between D2D and cellular networks, as an alternative radio interface with a different protocol is used for D2D communication. The out-band D2D communication is categorized into controlled and autonomous methods. In the controlled out-band D2D communication, a BS coordinates radio interfaces between devices, and DUEs have a pre-allocation on their spectrum resources. The BS can then meet the QoS requirement by prioritizing transmission for a particular user. The major setback of this method is the long delays, which are experienced due to the increase in network size that also increase the signalling overhead and reduce the performance of a network. To reduce the BS workload of an out-band D2D communication system, control of D2D communications can be transferred to the users' devices. This method of D2D communication is known as autonomous out-band D2D. The autonomous out-band is especially attractive for mobile service providers and operators since the devices in the D2D network control resource allocations among themselves, and hence the signalling overhead would be reduced. Moreover, the autonomous out-band eases the deployment of the BS as the network traffic management would be offloaded to the mobile devices.

Although reusing the available bandwidth in an underlay D2D communication system can significantly improve the spectrum efficiency and increase the overall capacity of the network [28], it requires careful consideration in allocating the available resources. This brings up the next challenge in D2D communications, which is known as resource management. Resource management deals with the efficient ways of managing resources to mitigate interference, save power and maximize capacity. The available resources that can be controlled are the transmitted power and the allocated spectrum. With the addition of D2D users to a cellular network, spectrum resource allocation is needed to be carefully managed for achieving the required level of QoS [29]. Power control is another aspect of resource management. This includes controlling

transmission power for both uplink (UL) and downlink (DL) of a cellular network as well as the transmission power of D2D communication transmitters.

Mobility management is also an essential aspect of D2D communications since DUEs may change their location while communicating with each other. As a result, their connectivity and communication can get interrupted. Mobility management consists of two operations, namely the location management and handoff management [30]. Location management enables the network to track the attachment points of mobile terminals between consecutive communications as they roam around a network [31]. Handoff management enables a network to maintain users' connections as a user moves from one attachment point to another.

Finally, D2D communication is vulnerable to the same security and privacy threats as cellular and ad-hoc wireless networks. Security threats can affect authentication, confidentiality, integrity, and availability of a network [32]. D2D communication requires efficient security solutions to enable secure, private, and trusted data exchange between devices. The security problem is needed to be dealt with in the concept of proximity-based direct communication and without any assistance from a cellular network.

In this research, we consider an underlaid D2D communication system, and the focus is on resource management to achieve a certain level of QoS in the system. The main aspect of resource management is to control the interference caused by D2D links. Therefore, in the next section, we will review the interference management techniques in D2D communication systems.

2.3. Interference Management in D2D Communication Systems

By integrating DUEs into a cellular network, the cellular architecture evolves into a two-tier cellular system, known as the macro and device tier. The macro tier includes the cellular communication between BS and CUEs, and the device tier involves D2D communications [33]. The system's design should be such that the implementation of the device tier does not affect the ongoing macro-tier communication services. The future cellular network will have to support various devices, including many low-power transmitters, which are part of its D2D communication network. Therefore, interference management in the future cellular network is

a non-trivial task [34]. The two main leverages that can be used to mitigate the effect of interferences are controlling the spectrum and power resources.

The cross-interferences between CUEs and DUEs can be alleviated by separating the operational spectrum of cellular and D2D communication links. The ultimate separation can be achieved in the out-band D2D, where separate bands are considered for D2D and cellular communications. However, in the out-band D2D, a cellular network does not have any control over other radio sources operating over the unlicensed spectrum. Therefore, such resources can cause uncontrollable interferences to the ongoing D2D communication links and reduce QoS. As a result, achieving the required QoS cannot be guaranteed in an out-band D2D communication system [35].

On the other hand, QoS management is feasible if all transmission sources become controllable within the same system. In the in-band D2D, either the UL or DL cellular spectrum is shared among D2D communication links. This makes the BS and cellular mobile devices the only transmitting sources, and so the interference level can effectively be managed within the cellular system. However, as a result of sharing a frequency spectrum, such a system is prone to intrasystem interference. The interference of the in-band D2D communication can be reduced by allocating a portion of the cellular spectrum specifically for D2D communication. The limitation of this method, which is known as the overlay D2D, occurs when the allocated cellular spectrum for D2D communications is used inefficiently, which leads to reduced resource utilization and spectrum deficiency [36].

The UL spectrum is always under-utilized, and there would be enough bandwidth to be reused for D2D links without affecting the performance of the cellular system. As a result, sharing the UL resources is the preferable choice in the underlay D2D communication systems [37]. However, reusing spectrum resources by the D2D links may cause significant degradation to the system performance. In the next section, the research that has been done on interference management in the underlaid communication systems will be reviewed.

2.3.1. Interference Management in Underlaid D2D Communication Systems

The available resource blocks in an underlaid D2D communication system are simultaneously used by cellular and D2D users. Therefore, the effect of interference caused by D2D links on the cellular links should be managed through techniques other than spectrum and time separation. Such techniques include power management and spatial-division multiplexing. The most common solution for controlling interference in an underlay D2D system is by restricting the transmission power between D2D links [38]. Power management in a UL underlaid D2D communication system should automatically be done by adjusting power levels in UEs during UL transmissions.

Due to the increasing demand for achieving higher throughputs in cellular networks, optimization of the power and spectrum utilization is a necessity for the overall improvement of the system capacity. Although increasing power in the transmitter would directly result in a capacity increase; it indirectly increases the interference level between devices that share the same spectrum. The power control algorithms can be employed in the context of resource management, which means combining the power control with mode selection to achieve a required system performance [39, 40].

The power control algorithms are categorized into centralized and distributed methods. In centralized algorithms, a BS controls and makes the necessary decisions for allocating power and spectrum resources. In contrast, in distributed algorithms, each mobile device decides independently on selecting the level of transmission power. Parameters such as maximum transmission power, CSI, number of resource blocks and received power per resource block are considered in different power control algorithms. It is common to implement power allocation in a centralized manner. However, the centralized schemes may not be suitable in a large network as they involve a high level of complexity. Instead, distributed power allocation techniques are preferable in such networks.

It was shown in [41] that a centralized power allocation method could maximize the sum rate of D2D links and guarantee reliable cellular communications. However, it requires knowledge of the global CSI at the BS, which increases the system's overhead. On the other hand, a distributed method that maximizes the D2D links' sum rate would not be able to guarantee the required QoS of the cellular links. Authors in [42] showed that in a centralized method, the maximum throughput could be achieved if a BS can control the mode of communication in D2D links. In other words, by allowing a BS to select the communication spectrum in D2D links, higher throughput can be obtained. However, in such a system, all devices should be capable of switching between the different operational spectrum.

A combination of spectrum and power allocation has been considered in [43] to maximize the overall network throughput while maintaining the required QoS. In this method, a BS only allows admission of D2D communications if the established D2D link has the potential to achieve the minimum QoS based on the value of the signal-to-interference-plus-noise ratio (SINR) and does not interfere with the ongoing cellular links spectrum. Therefore, a BS should continuously monitor the power and spectrum allocation throughout the network, which increases its workload. [44] proposed that the power control mechanism offload the monitoring workload to the D2D users. They assumed a system wherein a BS constantly transmits the maximum allowed power to the devices so that their transmission power does not exceed a certain threshold during D2D communications. Moreover, D2D users were only allowed to communicate during the uplink frame to reduce the interference effects on the cellular users.

Authors in [45] modelled an underlaid D2D communication system where the uplink spectrum could be used by the cellular and D2D users simultaneously and without any constraint. They then optimized the system based on the power transmitted by the D2D users to achieve the maximum rate and energy efficiency for the D2D links. However, no QoS for the communication links were considered. [46] modelled an underlay D2D system, where each D2D link can utilize multiple resource blocks from different cellular users. Based on this assumption, they proposed an optimum power allocation scheme to maximize the cell sum rate and D2D throughputs. In their model, the interference from the neighbouring cells was also considered. However, only the QoS for the cellular links were guaranteed, and the D2D links should transmit the maximum power accordingly. The utilization of resource blocks from multiple cellular users by a D2D pair is also considered in [47]. The main objective of this work was to allocate the power such that the QoS on both cellular and D2D links can be guaranteed. The system was modelled based on the statistical CSI, and multiple optimal and suboptimal power allocation methods were proposed. [48] considered the energy-efficiency and the QoS for both D2D and cellular links. In their work, they first maximized the energy efficiency of

D2D pairs one by one according to the required QoS, and then a spectrum allocation was done between the cellular and D2D users to minimize the interference. However, the complexity of this proposed algorithm is very high, which imposes a high workload on the BS. In [49], a more realistic system based on randomized D2D users' locations where considered. For such a system, power control and channel allocation methods based on the distances between D2D and cellular users were proposed, which resulted in decreasing the density of active D2D users sharing the same resources.

Recently, [50] studied the Simultaneous Wireless Information and Power Transfer (SWIPT) in D2D communication systems. Under this assumption, the interference can be exploited as beneficial signals to devices for energy harvesting. Therefore, the power allocation problem for achieving energy-efficiency can be treated differently as the transmitted power can be harvested by the other users.

Spatial-division multiplexing has also been investigated to reduce the interference level in underlaid D2D communication systems. Massive MIMO (Multiple-input Multiple-output), which is also known as large-scale antenna systems (LSAS), has been recognized as a promising technology for achieving higher data capacity in the next generation of mobile-cellular networks. Massive MIMO is mainly used for spatial-division multiplexing, where different data streams occupy the same frequencies and times. The concept of massive MIMO and spatial-division multiplexing can be used to develop an efficient multiuser MIMO (MU-MIMO) communication system.

In contrast to single-user or point-to-point MIMO, an MU-MIMO cellular communication system consists of a BS with multiple antennas for simultaneously servicing a set of multiple or single antenna users [51]. All users in such a system share multiplexing gains. An independently controllable array of antennas enables MU-MIMO systems to carry data on focused beams. In a cluttered propagation environment, data streams can be received from different directions simultaneously. A well-designed MU-MIMO system can reinforce data streams constructively where desired and interfere destructively where unwanted.

An MU-MIMO system requires knowing the frequency response of every user's propagation channel. Such information is obtained based on measurements rather than assuming channel characteristics. An LS-MIMO system uses a known training signal to estimate the frequency response of both UL and DL channels. Once the channels have been estimated, the CSI has to be utilized promptly before users' channel conditions significantly change. Hence, there is only a limited amount of time available for training. Both training and data transmission takes place in a time slot, which its duration is chosen so that no user can move more than a fraction of a wavelength during that time. CSI is used for both precoding in downlink and decoding in the uplink. Precoding includes mapping of data streams to drive each antenna for beamforming. An increasing number of antennas help to form beams, which can be focused more selectively on a particular user [52]. Decoding consists of designing a matched filter for the uplink data transmission.

Received symbols in MIMO communication systems are subjected to impurities, such as interference and random noise. These symbols can be detected separately or jointly. A joint detection has a better performance compared to a separate detection as characteristics of other symbols will be taken into account for detecting each symbol [53]. Co-channel interference (CCI) is an essential aspect of MIMO communication systems, which can offer essential benefits to MIMO communication detection techniques. Nevertheless, CCI enforces some limitations as well. For instance, CCI problem-solving algorithms have high computational complexity for optimum MIMO detection. Such complexities increase with the number of decision variables. However, with the advancement of computing power over the years, such limitations may no longer be regarded as a hold-up for practical applications. CCI's main feature is the overlapping of multiple transmissions. Such overlapping can take place over a frequency band or a time slot.

The interplay between MIMO systems and D2D communications can be employed to control the level of interference in an underlaid system as the D2D communication links cause continuous interference on the ongoing cellular communication transmission. It is shown in [54] that when the number of antennas at a BS is sufficiently larger than the number of serviced users, the users' channels can be designed to be orthogonal. Therefore, by employing a large array of antennas, the interference between D2D and cellular communications can be nullified. Such a technique allows close to zero interference on cellular links from underlaid D2D communication users. Although large arrays of antennas at a BS can assist in controlling the effect of interferences, by scaling the number of active users, D2D communication still contributes to the overall interferences of a cellular network through reusing radio resources. Therefore, to protect D2D links, the number of active D2D users has to be limited. The majority of existing research on D2D communication is based on single-antenna users, while research on large antenna arrays has just begun. However, the estimation of CSI in a cellular system with an underlaid D2D communication will be less accurate. This is a result of CCI presence during the channel estimation process and is one of the trade-offs in supporting D2D communication in a massive MIMO communication system.

The work of [55] introduced a power control solution in an underlaid D2D communication system benefiting from Massive MIMO to control the interference without a need to obtain the global CSI of the system. They showed that adding more antennas to the massive MIMO system can improve the spectral and energy efficiency of the whole system. However, a critical factor in such a system is the maximum number of D2D devices that can simultaneously be operational in the network. [56] studied the feasibility of pilot reuse by D2D users in an underlaid massive MIMO system to reduce the signalling overhead during CSI estimation. They introduced a pilot scheduling and power control algorithm in D2D devices to mitigate their effects on the cellular network channel estimation. Moreover, a power control algorithm has been proposed to maximize spectral efficiency while obtaining the required level of QoS. In [57], a power control method for a massive MIMO system with underlaid D2D communication links is proposed that jointly optimizes the power for both data and pilot signals.

2.4. Application of D2D Communications in Vehicular Communications

IEEE 802.11p [23] and cellular-based V2X communications are the technologies that are considered for vehicular communications. Until 2016, IEEE 802.11p, which is part of wireless access for vehicular environments (WAVE) protocol [58], was the only accepted standard for vehicular ad hoc networks (VANETs) [59]. However, there are two types of delays that are associated with the WAVE. One is as a result of the carrier sense multiple access with collision avoidance (CSMA/CA) mechanism used by the IEEE 802.11p protocol, and the other is due to the WAVE limited transmission range that is typically 100-200m [60]. Moreover, WAVE can only support a maximum of 800 Bytes in each transmission that happens every 100ms, and therefore, cannot guarantee the required QoS and access delay for many types of ADAS applications [61].

Cellular-based V2X was proposed by the 3GPP in Release 14 of LTE (LTE-V2V) [62] and has the potential to offer the required performances needed by ADAS applications [2, 61-64]. Moreover, it can benefit from the already available and almost ubiquitous coverage of cellular LTE systems. Fast channel variation and high mobility of users are parts of a vehicular communication environment. However, such environmental challenges can be addressed by employing direct and short-distance communication links. D2D communications enable nearby mobile devices to communicate with one another without their data traversing the core network or cellular BS and usually takes place over short distances. As a result, it can provide a low-latency, high data-rate transmission while spending a relatively small amount of power [65]. Such specifications are essential for an efficient vehicular communication system and make D2D communications suitable candidates for establishing V2V links between nearby vehicles [66].

The main links in vehicular communications are V2V and V2I, which are respectively supported by D2D and Device-to-BS links in a cellular-based V2X. It was shown in [67] that a better delay performance could be achieved by introducing overlay-D2D to a WAVE-based VANET. However, the limitation of overlay D2D occurs when the allocated cellular spectrum for D2D communication is inefficiently used, which leads to reduced resource utilization and spectrum deficiency. The spectrum deficiency problem can be mitigated by employing underlay D2D under careful resource management. In the next section, we review the works that have been done on employing underlaid D2D systems in vehicular communications. The main aspect of this research is on achieving the required QoS in communication links while mitigating the interference effects.

2.4.1. Interference Management in D2D-based Vehicular Communications

A centralized resource allocation for V2I and V2V connections in an underlay D2D communication system can be determined by a BS, assuming it poses the capability of choosing the best resource sharing scheme for V2V and cellular connections. V2V and V2I communication links are allowed to share the same resources for their individual data transmission. Such resource management is very crucial. D2D communication techniques for supporting V2V links require a more detailed study, as the rapid change of channel state has to be taken into account in radio resource management. A feasibility analysis has been carried out

in [68] to evaluate the applicability of the D2D underlay resource allocation for the support of joint V2V and V2I connections, and it was shown that D2D-assisted vehicle underlay communication can surpass traditional V2V techniques.

It was shown in [69] that the overall network throughput is not necessarily enhanced by employing more D2D transmissions between vehicles in an urban environment. This is because the SINR in the underlay D2D systems is sensitive to the number of D2D links. A vehicular communication network based on an underlay D2D communication system was envisaged in [70], where each D2D link could only share a spectrum with one cellular link at a time. Based on this assumption, different resource allocation methods were proposed to maximize the throughput of V2I links while maintaining the minimum QoS for V2V. [71] proposed a clusterbased system wherein multiple V2V links could share spectrum with a cellular link. The objective of this research was to allocate power for maximizing the sum rate of cellular users while satisfying the latency and reliability requirements of V2V links according to the vehicular communications requirements proposed by the European union METIS project [72]. In their method, both clusterings of V2V users and power allocation processes should be done at the BS. Power allocation based on machine learning techniques has been suggested in [73] for D2D-based V2V communications. This work aimed to improve the energy efficiency of the overall system while maintaining the QoS in V2V links. It was shown that after the training phase, the amount of the required computation would be small enough to be performed on the users' equipments. [74, 75] have optimized power allocation for D2D-based vehicular communication systems to meet the required QoS in V2V and V2I links considering different latencies related to the packets' transmission. They studied the probability of packet loss in the reliability of the system. The power optimization in their works aimed to achieve the maximum sum rate. The sharing of the spectrum by V2V users was studied in [76] by taking data security into account. They have also optimized their system to achieve the maximum sum rate. In [77], a suboptimal power allocation algorithm based on the game theory was proposed to maximize the minimum throughput of V2V links under the constraint of a minimum value for V2I links. Xiao et al. [78] considered energy harvesting in their system and, based on that, optimized the power allocation to obtain the maximum energy efficiency. Moreover, in contrast to other studies, they assumed an underlay D2D system by sharing the DL spectrum.

The main assumption in the abovementioned papers was that the slowly varying CSI of all links are perfectly known by the system and can be used in the resource allocation algorithms. Liu et al. [79] made a more realistic assumption, which was the knowledge of delayed CSI by the users. Based on this assumption, they proposed an optimized power allocation scheme to maximize the system's sum rate.

2.5. D2D Communications Performance Evaluation

Different objectives, such as power efficiency and spectral efficiency, have been considered for optimizing the performance of UL D2D underlay cellular systems. In [80], an underlay D2D cellular system with a fixed number of mobile devices was considered, and it was shown that by employing D2D, the amount of power consumption of the whole system was reduced. The authors of [81] considered a single-cell scenario that the BS is aware of the CSI and can control DUEs' transmit powers. They showed that in a system with one CUE and a pair of DUEs, the sum rate can be increased by up to 45%, provided that the system knows the interference of each link as a priori information.

QoS-aware communication systems are of great importance in situations that communication links are supposed to provide minimum transmission rates. [43] proposed a power allocation scheme to obtain a QoS-aware D2D system. However, the method was only limited to a situation that a pair of DUEs at its most shares spectrum with only one CUE. Authors in [44] suggested that DUEs should be able to adjust their transmission power level according to the required QoS by other CUEs and DUEs in the system. However, their work is based on the assumption that DUEs know the location and CSI of CUEs, which is either not feasible in a real system or imposes a high overhead signalling to the system.

2.6. Summary

In this chapter, the most recent publications on D2D communication and cellular-based vehicular communications were reviewed. D2D communications are considered as a solution for improving power and spectrum utilization in the next generation of cellular networks. D2D links use less power to transmit higher data rates with fewer delays between user equipments in a cellular system. However, with the addition of D2D users to a cellular network, spectrum

and power resources are needed to be carefully managed for achieving the required level of QoS. D2D communications are also suitable candidates for establishing V2V links between nearby vehicles. Studies have shown that D2D-assisted vehicular communication can surpass traditional V2V techniques. The next chapter discusses the mathematical concepts that are used for modelling D2D communication systems.

CHAPTER 3 : MATHEMATICAL MODELS

3.1. Introduction

In this chapter, we provide the mathematical concepts that will be used in the following chapters to model our proposed solution for power allocation in a D2D-based vehicular communication system. We first cover the cellular system that has been considered in our analysis. This follows by the model of the wireless channel. Finally, we present the mathematical formula for the LS-MIMO, which will be used in our considered underlaid D2D system.

3.2. Cellular Systems

A cellular system is an infrastructure-based wireless network, which operates based on the concept of reusing spectrum for spatially separated locations, known as a cell. A base-station, or an access point, is located at the centre of a cell and provides a centralized control mechanism for dynamic allocation of resources, controlling power, scheduling transmissions and handoff. Reusing of the spectrum is based on the fact that a base station's propagation power falls off with the distance, and therefore, in an ideal scenario, an area can be divided into non-overlapping cells. It is typical to consider a cellular network with non-overlapping hexagonal cells, where each cell neighbouring six other cells. Although in practical situations, circular shape cells are closer to the wave propagation model.

A base station uses the DL channel to send information to cellular users, and cellular users use the UL channel to communicate with a base station. In a cellular system, the UL and DL channels are separated into orthogonal signalling dimensions using time-division duplexing (TDD) and frequency-division duplexing (FDD). As, both, UL and DL channels have to be shared among different users in a cell, using multiple access techniques are required to divide the available signalling dimensions among the cellular users. Time-division multiple access (TDMA), Frequency-division multiple access (FDMA), Code-division multiple access (CDMA), Space-division multiple access (SDMA) and their hybrid combination are different multiple access methods used in cellular systems.

In the 5G cellular system, the length of each radio frame is 10 ms. Each frame is then divided into 10 subframes of length 1 ms. Subframes are formed by the number of slots, which varies according to the numerology used for the 5G, and each slot contains 14 OFDM symbols. The length of each slot is dependent on the 5G configuration and is equal to $\Delta T = \frac{1}{2\mu}$ ms, where $\mu = 0,1, ..., 4$ and represents the employed numerology. The 5G has the flexibility of assigning a consecutive number of symbols in each slot as DL, UL or Flexible. The combination of DL, UL and Flexible symbols in a slot is defined by the slot format. A resource block (RB) in the 5G cellular system is made up of one slot in the time domain and 12 sub-carrier in the frequency domain. The sub-carrier spacing is also dependent on the 5G configuration and is equal to $\Delta f = 2^{\mu} \times 15$ kHz. It is important to note that numerology 4 does not support data transmission and is only used for synchronization. The maximum supported bandwidth for $\mu = 0$ is 40 MHz, which can contain 270 RBs, and this for $\mu = 3$ is 400 MHz that supports 264 RBs. In the LTE cellular system, the number of slots per subframe is 2. So the length of each slot is 0.5 ms. The LTE bandwidth varies from 1.4 to 20 MHz, but the subcarrier spacing is always 15 KHz. Therefore, a 1.4 MHz LTE system contains 6 RBs and this for a 20 MHz is 100 RBs.

A cellular has to cope with interference from within a cell, known as intracell interference, as well as the interference from outside a cell, which is from neighbouring cells, referred to as co-channel or intercell interference. The co-channel interference is a result of channel reuse by the other cells, and the intracell interference is due to non-orthogonal channelization or the multipath effect and lack of effective synchronization. Nevertheless, in an underlaid D2D communication system, which will be explained in the coming sections, the co-channel interference is also considered as part of intracell interference as the spectrum resources are used by multiple D2D and cellular devices, simultaneously.

Both types of interferences would affect the amount of SINR at the receiver, which is calculated as

$$SINR = \frac{P_r}{N + P_I} \tag{3.1}$$

where P_r is the received signal power, N is the noise power and P_I is the received power caused by both intercell and intracell interference.

The performance of a cellular system depends on SINR, and mitigating interference is a key factor for improving the performance of a cellular system. Intercell interference can be controlled by increasing the distance between the cells using the same channels. On the other hand, intracell interference can be managed by spectral, temporal and spatial resource allocation, such as employing orthogonal signalling dimensions for multiple access as well as by careful power control mechanisms.

3.3. The Model of a Wireless Channel

Wireless multipath propagation is a phenomenon of a transmitted signal reaching a receiver by more than one propagation path. In an ideal broadcasting wireless system, the transmitted signal radiates in all directions from the transmitter to an infinite distance with no degradation. However, different types of attenuations will affect transmitted signals in practical systems. Such effects in a mobile radio propagation system are described in terms of large-scale and small-scale effects. Large-scale effects include the variation of the mean received signal strength over large distances, and small-scale effects involve the variation of the mean received signal strength over small distances relative to the signal wavelength. In addition to the smallscale and large-scale effects, the transmitted signals in a wireless medium will also be corrupted by random fluctuations of electrons within the air, which is known as noise. Channel models can also describe based on the empirical models and according to the measurements of the received signals.

The most common impairments factors for multipath degradation propagation are reflections, diffractions, and scattering from terrestrial environments such as buildings, mountains, trees, etc. Reflections occur when a radio signal collides with a massive object with large dimensions compared to the wavelength of the propagating signal. Diffractions happen when a surface obstructs the radio propagation path between the transmitter and the receiver with sharp edges. Scattering occurs when the radio signal travels through a medium consisting of objects with dimensions that are small compared to the signal wavelength. Generally, the

most critical large-scale impairments in a mobile channel are path loss and shadowing. Fading, on the other hand, is due to small-scale effects.

3.3.1. Path Loss and Shadowing

Free-space transmission between a transmitting antenna and a receiving antenna introduces a reduction in the transmitted signal power density. Path loss models are used to predict the transferred power and quantify the difference between the transmitted power, P_t , and received power, P_r . Numerous parameters are affecting the path loss of a mobile radio propagation beam. Base station antenna height, mobile antenna height and operating frequency are among such parameters, but overall the path loss is equal to

$$P_{L_{\rm dB}} = 10\log\frac{P_t}{P_r} \tag{3.2}$$

An accurate path loss model can be achieved from empirical models. However, a simple model for path loss can be useful to analyse various system designs. The simplified path loss model is given in [82] by

$$P_{L_{\rm dB}} = -A_{\rm dB} + 10\alpha \log d \tag{3.3}$$

where A_{dB} is a constant negative number depending on the antenna characteristics and α is known as the path loss exponent. Both of these values can be approximated from the empirical measurements. *d* is the distance between the transmitter and the receiver.

Shadowing is experienced due to the significant obstruction of the electromagnetic signal between the transmitter and the receiver. The shadowing effect experienced by the received signal power due to large buildings, trees, and hills between the transmitter and receiver propagation path. A log-normal model is a widely used model for the shadowing affected by a large number of shadowing objects. According to [82], in this model, P_t/P_r can be expressed as a random number with the probability density function (PDF) of
$$f(\psi_{\rm dB}) = \frac{1}{\sqrt{2\pi}\sigma_{\psi_{\rm dB}}} e^{\frac{-(\psi_{\rm dB} - \mu_{\psi_{\rm dB}})^2}{2\sigma_{\psi_{\rm dB}}^2}}$$
(3.4)

Where ψ_{dB} is the shadowing loss in dB and $\sigma_{\psi_{dB}}$ and $\mu_{\psi_{dB}}$ are the standard deviation and mean of ψ_{dB} log-normal random variable.

The path loss and shadowing models can be combined to model the power falloff versus distance. This combined model is given in [82] as

$$\left(\frac{P_r}{P_t}\right)_{\rm dB} = A_{\rm dB} - 10\alpha\log d - \psi_{\rm dB} \tag{3.5}$$

where ψ_{dB} is a zero-mean Gaussian random variable with the variance of $\sigma_{\psi_{dB}}^2$. Therefore,

$$\frac{P_r}{P_t} = \frac{A}{\psi} d^{-\alpha} = A\beta d^{-\alpha}$$
(3.6)

3.3.2. Small-Scale Fading Model

When a signal is affected by a large number of multipath components, statistical models must be used to model the channel. Multipath propagation leads to multipath fading, which exhibits randomness in the received signal strength, phase, and angle of arrival. As a result, the receiver sees the superposition of multiple transmitted copies, each of which experiencing attenuation, phase shift, and delay. According to [82], the baseband equivalent of a received signal through a fading channel is

$$r_{l}(t) = \int_{-\infty}^{\infty} h(\tau, t) s_{l}(t-\tau) e^{-j2\pi f_{c}\tau} d\tau$$
(3.7)

where f_c is the carrier frequency of the transmitted signal.

As can be seen, the signal is affected by the channel time spread, τ , and attenuation factor, h, which both are a function of time, t. Depending on the motion between a transmitter and a receiver, a multipath channel may be considered time-variant or time-invariant. The Doppler

spread, B_d , characterizes the time variation of a fading channel. In mobile communication, B_d is specified by the speed of a mobile receiver as

$$B_d = \frac{f_c}{c} \nu \tag{3.8}$$

where v (m/s) is the speed of the mobile, c is the speed of light (3e8 m/s) and f_c is the transmission carrier frequency. $B_d = 0$ corresponds to a static channel.

For a transmitted signal with a transmission rate of 1/T signal per second, the channel is considered as a slow-fading channel if $1/T \gg B_d$, which means for at least one signalling interval, the channel attenuation and phase shift are essentially fixed. Otherwise, the channel is called fast-fading. Moreover, the spread of a channel, τ , is characterised by the maximum path delay, T_m . A channel is considered as frequency-selective if $T \ll T_m$ and flat or frequencynonselective if $T \gg T_m$.

For a flat, slow-fading channel, when a large number of reflections occur at the receiver, based on the central limit theorem, the multipath channel can statistically be modelled as a complex Gaussian random variable. For such a channel, each transmitted signal is corrupted by attenuation and a phase shift. If no line-of-sight (LoS) path exists between transmitter and receiver, the attenuation is modelled as a Rayleigh random variable, and if an LoS path exists, the attenuation has a Rician distribution. The phase shift has a uniform distribution in the range of $(0,2\pi)$. For an attenuation in the Rayleigh channel, the Probability Density Function (PDF) of |h| is

$$P(|h|) = \frac{2|h|}{\Omega} e^{-\frac{|h|^2}{\Omega}}$$
(3.9)

where $E\{|\mathbf{h}|^2\} = \Omega$ is the mean power of the received signal with the transmitted power of 1 watt, the PDF of the envelope of a received signal in a Rayleigh fading channel with $\Omega = 1$ is shown in Figure 3.1. The channel is in adverse condition with a probability of 0.6325.



Figure 3.1. PDF of the received signal envelope in a Rayleigh fading channel ($\Omega = 1$)

3.3.3. Noise

The most widely spread form of noise is thermal noise, which is due to thermal agitation of electrons and uniformly distributed across the frequency spectrum and mostly modelled using additive white Gaussian noise model. According to [83], the thermal noise gets generated by random fluctuations of electrons within the conducting media and has a Gaussian distribution with zero mean. From the Nyquist formula, an assumption is made of a resistor to contain a voltage generator across the terminals. The mean square value is

$$v^2 = 4RkT\Delta f \tag{3.10}$$

where R is the resistance of the conductor, k is the Boltzmann's constant, T is the absolute temperature and Δf is the signal bandwidth. The average power in a conductor due to thermal noise is equal to

$$P_{thermal} = kTB \tag{3.11}$$

Additive white Gaussian noise (AWGN) is a statistically random radio noise characterized by a Gaussian PDF. AWGN is the most used channel model for a communication system.

AWGN is modelled by adding a white Gaussian noise process to the transmitted signal. The received signal is mathematically expressed as

$$r(t) = s(t) + n(t)$$
 (3.12)

where r(t) and s(t) are the received and transmitted signals and n(t) is a real zero-mean white Gaussian process with single-sided Power Spectral Density (PSD) of N_0 (W/Hz).

3.4. Massive MIMO

The use of multiple antenna systems is a standard method to overcome fading degradation and combat interference effects. Employing a MIMO system can also help to multiply the capacity of a communication system. Point-to-point MIMO represents the purest form of MIMO, where the transmitter and the receiver are equipped with N_t and N_r antennas, respectively. An alternative approach for employing multiple antennas is the multi-user MIMO (MU-MIMO) system. In an MU-MIMO system, a single BS serves multiple users using the same time-frequency resource. A key component in MU-MIMO systems is the CSI knowledge of the UL channel at the BS. The CSI information is used in the pre-coder design for data transmission in DL and for a linear decoder in detecting the UL transmitted signal at the BS.

Let us assume an MU-MIMO system with K users and an M-antennas BS. In a UL transmission where all users transmit signals over the same time-frequency resources simultaneously, the received vector at the BS is

$$\boldsymbol{r} = \mathbf{G}\boldsymbol{s} + \boldsymbol{n} \tag{3.13}$$

where $s \in \mathbb{C}^{K \times 1}$ and $r \in \mathbb{C}^{M \times 1}$ are the vectors for the transmitted and received signals. $\mathbf{G} \in \mathbb{C}^{M \times K}$ is the transfer channel matrix of the system containing both large and small-scale effects as

$$\mathbf{G} = \mathbf{H}\mathbf{D}^{1/2} \tag{3.14}$$

 $\mathbf{H} \in \mathbb{C}^{M \times K}$ denotes the small-scale fading effects modelled by i.i.d. $\mathcal{CN}(0,1)$ elements and $\mathbf{D} \in \mathbb{C}^{K \times K}$ is a diagonal matrix to model the large-scale impairments (i.e., path loss and shadowing). Finally, $\mathbf{n} \in \mathbb{C}^{M \times 1}$ represents the AWGN vector modelled by i.i.d. $\mathcal{CN}(0, N_0)$ elements. \mathbb{C} is the field of complex numbers, and it is assumed that \mathbf{H}, \mathbf{D} and \mathbf{n} are statistically independent. The detection task in a MIMO system is to estimate \mathbf{s} based on the knowledge of \mathbf{r} . If \mathbf{G} is known at the receiver as a result of CSI, the detection is coherent; otherwise, it is noncoherent.

In a favourable propagation, it is assumed that the small-scale attenuation of different channels are independent and therefore, the following property will be held theoretically when $M \rightarrow \infty$ and in practice $M \gg K$ [84]. The proof is based on the law of large numbers and can be found in [84].

$$\frac{\mathbf{G}^{\mathbf{H}}\mathbf{G}}{M} = \mathbf{D}^{1/2} \frac{\mathbf{H}^{\mathbf{H}}\mathbf{H}}{M} \mathbf{D}^{1/2} \approx \mathbf{D}$$
(3.15)

Such a system with a larger number of antennas at the BS than the active users is known as the massive MIMO or Large-scale MIMO (LS-MIMO) system. In an LS-MIMO system, M is typically much more significant than K. Moreover, CSI is only required at the BS. A large number of antennas make the beamforming gains constant over frequency. Therefore, the received signal is only affected by the large-scale fading coefficients, which themselves are independent of frequency and antenna index.

In this research, a Zero-Forcing (ZF) detector under a perfect channel estimation at the BS is considered to detect the received UL signals. The ZF filter is equal to

$$A = \mathbf{G}(\mathbf{G}^{\mathbf{H}}\mathbf{G})^{-1} \tag{3.16}$$

and the received signal, r, can then be separated into a vector $y \in \mathbb{C}^{K \times 1}$ as

$$y = A^{H}r = s + (G^{H}G)^{-1}G^{H}n = s + D^{-1/2}(H^{H}H)^{-1}H^{H}n$$
(3.17)

We denote the signal transmitted by the k^{th} user, which is the k^{th} element of vector s, as $[s]_k = s_k$. We further assume the average transmitted power of each user to be P, and so $\mathbb{E}\{|s_k|^2\} = P$ for k = 1, ..., K. The average noise power on the k^{th} stream of the vector y under the favourable propagation constraint can be calculated as

$$\mathbb{E}\{[(\mathbf{G}^{\mathbf{H}}\mathbf{G})^{-1}\mathbf{G}^{\mathbf{H}}\boldsymbol{n}]_{kk}\} = \mathbb{E}\{D_{k}^{-1}(\boldsymbol{h}_{k}^{H}\boldsymbol{h}_{k})^{-1}|n_{k}|^{2}\}$$
$$= D_{k}^{-1}\mathbb{E}\{(\boldsymbol{h}_{k}^{H}\boldsymbol{h}_{k})^{-1}\}N_{0}$$
K

$$= \frac{\mathbb{E}\{tr[(\mathbf{H}^{\mathbf{H}}\mathbf{H})^{-1}]\}}{KD_{k}}N_{0} = \frac{\frac{K}{(M-K)}}{KD_{k}}N_{0} = \frac{1}{(M-K)D_{k}}N_{0}$$
(3.18)

where $[D]_{kk} = D_k$. Therefore, the SINR at the BS for the k^{th} user is

$$SINR_k = (M - K)D_k \frac{P}{N_0}$$
(3.19)

The ergodic spectral efficiency of the k^{th} user is then equal to

$$\mathbf{R}_{k} = \mathbb{E}\left\{\log\left(1 + \frac{P}{\left[(\mathbf{G}^{\mathsf{H}}\mathbf{G})^{-1}\right]_{kk}N_{0}}\right)\right\} \ge \log\left(1 + \frac{P}{\mathbb{E}\left\{\left[(\mathbf{G}^{\mathsf{H}}\mathbf{G})^{-1}\right]_{kk}\right\}N_{0}}\right)$$
(3.20)

Where the inequality is based on Jensen's inequality. By substituting (3.19) in (3.20) we get

$$R_k \ge \log\left(1 + (M - K)D_k \frac{P}{N_0}\right)$$
(3.21)

As is seen from (3.17) and (3.21), by increasing the number of antennas, the fast fading effects can be reduced and theoretically vanishes when the number of antennas approaches infinity and the spectral efficiency of the user's increases.

3.5. Summary

In this chapter, the required mathematical background for modelling a D2D communication system, which will be covered in the next chapter, was discussed. This includes different types of impairments that affect a wireless communication system. These impairments are categorized into two groups as large-scale and small-scale effects. Large-scale effects include the variation of the mean received signal strength over large distances, and small-scale effects involve the variation of the mean received signal strength over small distances or short time intervals. The most critical large-scale impairments in a mobile channel are path loss and shadowing. Fast fading, on the other hand, is due to small-scale effects. The use of multiple antenna systems, which is a standard method to overcome fading degradation and combat interference effects, was also discussed. Although different types of filters can be used as the receiving filter, ZF filters can achieve satisfactory performance in cancelling interference in an MU-MIMO system with a large number of antennas.

Chapter 4 : POWER ALLOCATION FOR A QOS-AWARE UNDERLAY D2D COMMUNICATION SYSTEM

4.1. Introduction

The implementation of D2D communication must not affect nearby ongoing communication services. Therefore, resource management is a significant aspect of D2D communication, which aims to reduce interferences. Spectrum and power resources are the primary means that can be employed by interference control mechanisms. By separating the operational spectrum of cellular and D2D communications, the cross-interferences between CUEs and DUEs can be mitigated. Alternatively, when DUEs and CUEs share a spectrum, the interference effects can be limited by careful power allocation.

In this chapter, we study an underlay D2D communication in a cellular system, where the cellular UL spectrum is shared among DUEs and CUEs in a random competitive manner. By strictly controlling the maximum transmission power, DUEs can reuse the UL spectrum without degrading the performance of CUEs communications. The intra-cell cross-interference between DUEs and CUEs in such a system is illustrated in Figure 4.1.



Figure 4.1. Illustration of intra-cell interference for an in-band UL D2D communication system.

In this chapter, we employ the statistical methods for analysing the performance of a D2D communication that underlays the UL of a cellular system. By sharing the UL spectrum for D2D communication, the system can benefit from employing multiple arrays of antennas to control the interference at the BS. We then derive the equations from calculating the minimum transmission power that provides the required spectral efficiency for both CUEs and DUEs, without considering any limitation on the spectrum sharing among devices. As the preference of D2D is for short-distance communications, we only consider D2D links between users that are in each other's LoS.

4.2. System Model

We investigate an underlay D2D communications that share the UL spectrum in a cellular network as in Figure 4.2. This is a standard practice because the uplink resources are less intensively used and CUE interference cancellation becomes more manageable by employing a very large antenna array at the BS. Each mobile device is equipped with a single antenna radio equipment and the BS receiver has *M* antennas.

DUEs directly communicate with each other in a distributed ad-hoc fashion with no BS assistance, and all mobile users are uniformly distributed over the cell coverage. We only consider intra-cell interferences. Furthermore, the specific modulation and coding schemes were not considered, and only the signals' power levels were considered.

The set of mobile devices involved in cellular communication in a cell are denoted by C = 1, 2, ..., N and the pair of devices involved in D2D communication as D = 1, 2, ..., O. We assume a fully-loaded network, where the active CUEs occupy all N orthogonal channels in the cell, and there is no spare spectrum. As the BS contains M antenna, the baseband received signal at the BS, $\mathbf{r}^{(c)} \in \mathbb{C}^{M \times 1}$, is equal to

$$\boldsymbol{r}^{(c)} = \sum_{k \in \mathcal{C}} \sqrt{A^{(c)} \beta_k^{(c)}} (d_k)^{-\frac{\alpha_c}{2}} \boldsymbol{h}_k u_k^{(c)} + \sum_{i \in \mathcal{D}} \sqrt{A^{(c)} \beta_i^{(c)}} (d_i)^{-\frac{\alpha_c}{2}} \boldsymbol{h}_i u_i^{(d)} + \boldsymbol{n}$$
(4.1)



Figure 4.2. A D2D underlaid massive MIMO system consisting of both CUEs and DUEs [7].

The large-scale attenuation effect consists of shadowing and path loss. Following the models provided in chapter 3, the path loss would affect the transmitted signal's magnitude by a factor of

$$\sqrt{A^{(c)}}(d)^{-\frac{\alpha_c}{2}} \tag{4.2}$$

where *d* denotes the distance between a mobile device (either CUE or DUE) and BS, $\alpha_c > 2$ is the path loss exponent, and $A^{(c)}$ is the path loss constant that depends on the antenna height and carrier frequency. Also, as a result of the shadowing, the signal magnitude is affected by a factor of

$$\sqrt{\beta^{(c)}} \tag{4.3}$$

where $\beta^{(c)}$ is modelled as i.i.d. log-normal shadow fading random variables with $10 \log \beta^{(c)} \sim \mathcal{N}(0, \sigma_{sc}^2)$.

The effects of small-scale fading are denoted by $\mathbf{h} \in \mathbb{C}^{M \times 1}$, and its elements are taken as i.i.d. complex Gaussian random variable distributed by $\mathcal{CN}(0,1)$. \mathbf{n} denotes additive white Gaussian noise (AWGN) vector, which its elements are i.i.d. random variables distributed by $\mathcal{CN}(0, N_0)$. $u^{(c)}$ and $u^{(d)}$ are the transmitted signals by the CUEs and DUEs, respectively. They are modelled as i.i.d. random variables with $\mathbb{E}\left\{\left|u^{(c)}\right|^2\right\} = P_c$ and $\mathbb{E}\left\{\left|u^{(d)}\right|^2\right\} = P_d$.

The received signal at the i^{th} DUE's receiver is

$$r_{i}^{(d)} = \sqrt{A^{(d)}} (d_{ii})^{-\frac{\alpha_{d}}{2}} g_{ii} u_{i}^{(d)} + \sum_{k \in \mathcal{C}} \sqrt{A^{(d)} \beta_{k}^{(d)}} (d_{ik})^{-\frac{\alpha_{d}}{2}} g_{ik} u_{k}^{(c)} + \sum_{\substack{j \in \mathcal{D} \\ j \neq i}} \sqrt{A^{(d)} \beta_{j}^{(d)}} (d_{ij})^{-\frac{\alpha_{d}}{2}} g_{ij} u_{j}^{(d)} + n \quad (4.4)$$

where d_{ik} is the distance between mobile device *i* and mobile device *k*, $\alpha_d > 2$ denotes the path loss exponent between two devices, and $\beta^{(d)}$ are i.i.d. log-normal shadow fading random variables distributed as $10 \log \beta^{(d)} \sim \mathcal{N}(0, \sigma_{sd}^2)$.

The effects of small-scale fading are denoted by g, which are i.i.d. $\mathcal{CN}(0,1)$ variables. n is due to AWGN and has complex Gaussian distribution of $\mathcal{CN}(0, N_0)$. We assume $u^{(c)}$ and $u^{(d)}$ are i.i.d. random variables where $\mathbb{E}\left\{\left|u^{(c)}\right|^2\right\} = P_c$ and $\mathbb{E}\left\{\left|u^{(d)}\right|^2\right\} = P_d$.

Let us assume $w_m^{(c)}$ is the filter used by the BS for receiving the signal from the *m*th CUE transmitter. Therefore, the averaged received SINR at the BS from the *m*th CUE link can be expressed as

$$\gamma_m^{(c)} = \frac{S_m^{(c)}}{I_m^{(c\to c)} + I_m^{(d\to c)} + \left\| \boldsymbol{w}_m^{(c)^*} \right\|^2 N_0}$$
(4.5)

where

$$S_m^{(c)} = P_c A^{(c)} \beta_m^{(c)} (d_m)^{-\alpha_c} \left\| \boldsymbol{w}_m^{(c)^*} \boldsymbol{h}_m \right\|^2$$
(4.6)

is desired signal power

$$I_{m}^{(c \to c)} = \sum_{\substack{k \in \mathcal{C} \\ k \neq m}} P_{c} A^{(c)} \beta_{k}^{(c)} (d_{k})^{-\alpha_{c}} \left\| \boldsymbol{w}_{m}^{(c)^{*}} \boldsymbol{h}_{k} \right\|^{2}$$
(4.7)

is the interference as a result of CUEs other than mth CUE and

$$I_{m}^{(d \to c)} = \sum_{i \in \mathcal{D}} P_{d} A^{(c)} \beta_{i}^{(c)} (d_{i})^{-\alpha_{c}} \left\| \boldsymbol{w}_{m}^{(c)^{*}} \boldsymbol{h}_{i} \right\|^{2}$$
(4.8)

is the interference caused by DUEs. Similarly, the received SINR at the *m*th DUE receiver can be expressed as

$$\gamma_m^{(d)} = \frac{S_m^{(d)}}{I_m^{(c \to d)} + I_m^{(d \to d)} + \left\| w_m^{(d)^*} \right\|^2 N_0}$$
(4.9)

where

$$S_m^{(d)} = P_d A^{(d)} \beta_m^{(d)} (d_{mm})^{-\alpha_d} \left\| w_{mk}^{(d)*} g_{mm} \right\|^2$$
(4.10)

is the desired signal power,

$$I_m^{(c \to d)} = \sum_{k \in \mathcal{C}} P_c A^{(d)} \beta_k^{(d)} (d_{mk})^{-\alpha_d} \left\| w_{mk}^{(d)^*} g_{mk} \right\|^2$$
(4.11)

is the interference as a result of CUEs and

$$I_{m}^{(d \to d)} = \sum_{\substack{i \in \mathcal{D} \\ i \neq m}} P_{d} A^{(d)} \beta_{i}^{(d)} (d_{mi})^{-\alpha_{d}} \left\| w_{mk}^{(d)^{*}} g_{mi} \right\|^{2}$$
(4.12)

is the interference caused by other D2D links.

The ergodic spectrum efficiency of the cellular link for the *m*th CUE is

$$R_m^{(c)} = \mathbb{E}\left\{\log\left(1 + \gamma_m^{(c)}\right)\right\}$$
(4.13)

Moreover, the ergodic spectrum efficiency of the *m*th DUEs pair is

$$R_m^{(d)} = \mathbb{E}\left\{\log\left(1 + \gamma_m^{(d)}\right)\right\}$$
(4.14)

For achieving the ergodic spectrum efficiencies of the links, no prior knowledge about the location of the users or the CSI is required.

4.3. Power Allocation

In many applications, achieving a minimum QoS requirement for DUEs and CUEs are essential. As it is seen from (4.13) and (4.14), the minimum QoS is directly proportional to $\gamma_m^{(c)}$ and $\gamma_m^{(d)}$ for the *m*th CUE and DUE, respectively. However, it can be deduced from (4.9) that if DUEs control their power entirely, a CUE can achieve the required QoS. This is mutually applicable to achieving a required performance by DUE as a result of strict control over CUE transmission power.

Employing a BS with *M* antennas offers a significant opportunity for achieving the desired QoS in CUE links. It is shown in [54] that if a BS contains a large array of antennas $(M \rightarrow \infty)$ and has the capability of obtaining the perfect CSI through the transmission of training sequences, then by designing $w_m^{(c)}$ as a ZF filter, $\gamma_m^{(c)}$ can increase unboundedly and therefore, any desired capacity is achievable.

However, having a perfect CSI in an environment that contains high mobility among users is not practical. To overcome this limitation, a training sequence needs to be transmitted every few hundred milliseconds for CSI estimation. The CUE spectral efficiency of a system that has a BS with a large array of antennas and an estimated CSI based on training sequences of length T_c ($T_c > N$) for N cellular users is achieved according to [54] as

$$R_{m}^{(c)} = \mathbb{E}\left\{ \log\left(1 + \frac{T_{c}P_{c}^{2}\left(A^{(c)}\beta_{m}^{(c)}\right)^{2}(d_{m})^{-2\alpha_{c}}}{\sum_{i\in\mathcal{D}}P_{d}^{2}\left(A^{(c)}\beta_{i}^{(c)}\right)^{2}(d_{i})^{-2\alpha_{c}}}\right)\right\} \geq \log\left(1 + \frac{T_{c}P_{c}^{2}}{\mathcal{D}P_{d}^{2}}\right)$$
(4.15)

Therefore, the minimum capacity requirement of CUE links, $R_0^{(\text{cue})}$, can be used to calculate the required transmission power, P_c and P_d , such that $R_m^{(c)} \ge R_0^{(\text{cue})}$ for all m.

The preference of D2D is for short-distance communications. Therefore, we consider the D2D links between DUEs that are in each other's LoS. The D2D spectral efficiency given in (4.14) for this system can be calculated based on $S_m^{(d)} = P_d A^{(d)} (d_{mm})^{-\alpha_d}$. The minimum capacity requirement of the D2D links is then equal to $R_0^{(due)}$ and $R_m^{(d)} \ge R_0^{(due)}$.

By considering that $\gamma_m^{(d)}$ is a convex function of $\{\beta_i^{(d)}\}, \{g_{mi}\}$ and $\{d_{ii}\}$ we have

$$R_m^{(d)} = \mathbb{E}\left\{\log\left(1+\gamma_m^{(d)}\right)\right\} \ge \log\left(1+\frac{P_d}{\mathbb{E}\left\{\beta_i^{(d)}\right\}(NP_c+(0-1)P_d)+\frac{N_0}{A^{(d)}}(\mathbb{E}\{d_{mi}\})^{\alpha_d}}\right)$$

$$\geq \log \left(1 + \frac{P_d}{(NP_c + (0-1)P_d) + \frac{N_0}{A^{(d)}} (\mathbb{E}\{d_{mi}\})^{\alpha_d}} \right)$$
(4.16)

The convexity of $\log\left(1+\frac{1}{x}\right)$ can easily be proved as $\frac{d^2}{dx^2}\log\left(1+\frac{1}{x}\right) = \frac{1}{\ln 2}\frac{2x+1}{(x^2+x)^2} > 0$ and the convexity of $\frac{1}{1+x}$ is proved by $\frac{d^2}{dx^2}\frac{1}{1+x} = \frac{2}{(1+x)^3} > 0$ knowing that x > 0.

We model the spatial positions of the user equipments as a Poisson point process (PPP). Therefore, for a user equipment m located at (x_0, y_0) and user equipment i located at (x_1, y_1) we have

$$d_{mi} = \sqrt{(x_0 - x_1)^2 + (y_0 - y_1)^2} = \sqrt{(r_0 - r_1 \cos \theta)^2 + (r_1 \sin \theta)^2}$$
$$= \sqrt{r_0^2 + r_1^2 - 2r_1 r_0 \cos \theta}$$
(4.17)

Without loss of generality, we can assume user equipment m angular position is at zero. For a circular cell with the coverage radius of R, r_0 and r_1 are uniformly distributed in the range of [0, R]. Moreover, θ is uniformly distributed in the range of $[0, \pi]$. The expected value for the distance between two randomly located user equipments are achieved from [85] as

$$\mathbb{E}\{d_{mi}\} = \frac{1}{\pi R^2} \iiint_{r_0, r_1, \theta} \sqrt{r_0^2 + r_1^2 - 2r_1 r_0 \cos\theta} \, dr_0 dr_1 d\theta = \frac{128}{45\pi} R \tag{4.18}$$

By simultaneously solving (4.15) and (4.16), the minimum required P_d and P_c will then be achieved as

$$P_{d} = \frac{\frac{N_{0}}{A^{(d)}} \left(\frac{128}{45\pi}R\right)^{\alpha_{d}}}{\left(2^{R_{0}^{(due)}} - 1\right)^{-1} - (0 - 1) - N\sqrt{(0 - 1)\left(2^{R_{0}^{(cue)}} - 1\right)}}$$
(4.19)

$$P_c = \sqrt{(0-1)\left(2^{R_0^{(\text{cue})}} - 1\right)} P_d \tag{4.20}$$

that gives us the required transmission power for DUEs and CUEs to achieve the desired transmission rate in an average statistical sense.

4.4. Summary

In this chapter, we have investigated an underlay D2D communications that share the UL spectrum of a cellular network, which benefits from a large array of the antenna at the BS. The concept of massive MIMO has been considered to minimise the interference in the user-to-cellular link. We have employed a statistical method to analyse the performance of the system without considering any prior knowledge about the location of the users, the CSI, or any limitation on the spectrum sharing among devices. We achieved the required power in a QoS-aware system by calculating the minimum transmission power that will provide the required spectral efficiency for both CUEs and DUEs.

Chapter 5 : POWER ALLOCATION FOR A D2D-ENABLED VEHICULAR COMMUNICATION SYSTEM

5.1. Introduction

Cellular-based V2X was proposed by the 3GPP in Release 14 of LTE (LTE-V2V) and had the potential to offer the required performances needed by ADAS applications. Moreover, it can benefit from the already available and almost ubiquitous coverage of cellular LTE systems. D2D communication enables the nearby mobile devices to communicate with one another without the data traversing the core network or a cellular BS, and therefore, can be considered as a promising solution for reliable and efficient V2V links. Furthermore, the V2I links can be supported by Device-to-eNodeB links in a cellular-based V2X.

In this chapter, a cellular-based vehicular communication system is considered that employs underlay D2D for establishing V2V links between vehicles. The complete uplink spectrum of a cellular system is shared with the underlay D2D communication system for this purpose. All vehicles can be involved in V2V or V2I communications without any limitation, but each V2V link only shares the spectrum with one V2I link at a time. We have employed the achieved results from the previous chapter to analyse the performance of the system without considering any prior knowledge about the location of the vehicles or the CSI.

This chapter aims to find boundaries on the allocated transmission powers to achieve sufficient spectral efficiencies for V2I and V2V links. For this purpose, we use the concept of massive MIMO to cancel the interference in the V2I links. We then derive the equations for calculating the minimum required transmission power to establish each link. As the preference of ADAS applications is to monitor a vehicle's surrounding environment for detecting hazardous conditions, we consider the V2V links between vehicles that are in each other's LoS.

5.2. System Model

Consider a cellular-based VANET shown in Figure 5.1. The V2V connections are supported by underlay D2D communication, which reuses the cellular uplink spectrum. This is a standard practice because the uplink resources are less intensively used and V2I interference cancellation becomes more manageable by employing a very large antenna array at the BS, as was mention in the previous chapter. Each vehicle is equipped with a single antenna radio equipment, and the BS receiver has M antennas. Therefore, a vehicle can either be involved in V2I or V2V communication at a time. We only consider intra-cell interferences in this chapter.



Figure 5.1. A vehicular communication network based on that employed underlay D2D sharing the cellular uplink spectrum.

We denote the set of vehicles in a cell that is involved in V2I communication by $\mathcal{I}=1,2,...,I$ and those that are involved in V2V communication by $\mathcal{V}=1,2,...,V$. Figure 5.1 illustrates a V2V link between the pair $j \in \mathcal{V}$ and a V2I link between the vehicle $i \in \mathcal{I}$ and the BS.

The M×1 dimensional baseband received signal at the BS, which is as a result of V2I communication is

$$\boldsymbol{r}^{(c)} = \sum_{i \in \mathcal{I}} \sqrt{P_c A^{(c)} \beta_i^{(c)}} (d_i)^{-\frac{\alpha_c}{2}} \boldsymbol{h}_i^{(c)} u_i^{(c)} + \sum_{j \in \mathcal{V}} \sqrt{P_d A^{(c)} \beta_j^{(c)}} (d_j)^{-\frac{\alpha_c}{2}} \boldsymbol{h}_j^{(d)} u_j^{(d)} + \boldsymbol{n}^{(c)}$$
(5.1)

where P_c and P_d are the V2I and V2V transmission power, respectively. The large-scale attenuation effect consists of shadowing and path loss.

Similar to the previous chapter, the Path loss is equal to $\left\{\sqrt{A^{(c)}}(d_i)^{-\frac{\alpha_c}{2}}\right\}$, where d_i denotes the distance between vehicle *i* and BS. $\alpha_c > 2$ is the path loss exponent, and $A^{(c)}$ is the path loss constant depending on the antenna height and carrier frequency. The shadowing effect is modelled by $\left\{\sqrt{\beta_i^{(c)}}\right\}$, where $\left\{\beta_i^{(c)}\right\}$ are i.i.d. log-normal shadow fading random variables with $10 \log \beta_i^{(c)} \sim \mathcal{N}(0, \sigma_{sc}^2)$.

The effects of small-scale fading are denoted by $\boldsymbol{h}_{i}^{(c)}$ and $\boldsymbol{h}_{j}^{(d)}$, where the elements of $\boldsymbol{h}_{i}^{(c)}$ and $\boldsymbol{h}_{j}^{(d)}$ are taken as i.i.d. complex Gaussian random variable distributed by $\mathcal{CN}(0,1)$. $\boldsymbol{n}^{(c)}$ denotes additive white Gaussian noise (AWGN) vector, which its elements are i.i.d. random variables distributed by $\mathcal{CN}(0, N_0)$.

The received signal at the receiving vehicle of the k^{th} pair involved in V2V communication is

$$r_{k}^{(d)} = \sum_{i \in \mathcal{I}} \sqrt{P_{c} A^{(d)} \beta_{i}^{(d)}} (d_{ik})^{-\frac{\alpha_{d}}{2}} g_{ik}^{(d)} u_{i}^{(c)} + \sum_{j \in \mathcal{V}} \sqrt{P_{d} A^{(d)} \beta_{j}^{(d)}} (d_{jk})^{-\frac{\alpha_{d}}{2}} g_{jk}^{(d)} u_{j}^{(d)} + n^{(d)}$$
(5.2)

where d_{ik} is the distance between the transmitting vehicle *i* and the receiver vehicle of the *k*th pair. $\alpha_d > 2$ denotes the path loss exponent, and $\{\beta_i^{(d)}\}$ are i.i.d. log-normal shadow fading random variables distributed as $10 \log \beta_i^{(d)} \sim \mathcal{N}(0, \sigma_{sd}^2)$.

The effects of small-scale fading are denoted by $g_{ik}^{(d)}$, which are i.i.d. $\mathcal{CN}(0,1)$ variables. $n^{(d)}$ is AWGN with complex Gaussian distribution of $\mathcal{CN}(0, N_0)$. For this paper, we assume $\{u_i^{(c)}\}, \{u_j^{(d)}\}$ are i.i.d. random variables distributed by $\mathcal{CN}(0,1)$.

Let us assume $\boldsymbol{w}_m^{(c)}$ is the filter used by the BS for receiving the V2I signal from the *m*th vehicle transmitter. Therefore, the received SINRs at the BS from the *m*th vehicle V2I link can be expressed as

$$\gamma_m^{(c)} = \frac{S_m^{(c)}}{I_m^{(c\to c)} + I_m^{(d\to c)} + \left\| \boldsymbol{w}_m^{(c)^*} \right\|^2 N_0}$$
(5.3)

$$S_m^{(c)} = P_c A^{(c)} \beta_m^{(c)} (d_m)^{-\alpha_c} \left\| \boldsymbol{w}_m^{(c)^*} \boldsymbol{h}_m^{(c)} \right\|^2$$
(5.3a)

$$I_{m}^{(c \to c)} = \sum_{\substack{i \in \mathcal{I} \\ i \neq m}} P_{c} A^{(c)} \beta_{i}^{(c)} (d_{i})^{-\alpha_{c}} \left\| \boldsymbol{w}_{m}^{(c)^{*}} \boldsymbol{h}_{i}^{(c)} \right\|^{2}$$
(5.3b)

$$I_{m}^{(d \to c)} = \sum_{j \in \mathcal{V}} P_{d} A^{(c)} \beta_{j}^{(c)} (d_{j})^{-\alpha_{c}} \left\| \boldsymbol{w}_{m}^{(c)^{*}} \boldsymbol{h}_{j}^{(d)} \right\|^{2}$$
(5.3c)

Where (5.3a) gives the desired signal power, (5.3b) is the interference as a result of V2I from vehicles other than *m*th vehicle, and (5.3c) is the interference caused by V2V links. Similarly, the received SINRs at the receiving vehicle of the *m*th pair involved in V2V communication can be expressed as

$$\gamma_m^{(d)} = \frac{S_{mm}^{(d)}}{I_m^{(c \to d)} + I_m^{(d \to d)} + N_0}$$
(5.4)

$$S_{mm}^{(d)} = P_d A^{(d)} \beta_m^{(d)} (d_{mm})^{-\alpha_d} \left\| g_{mm}^{(d)} \right\|^2$$
(5.4a)

$$I_m^{(c \to d)} = \sum_{i \in \mathcal{I}} P_c A^{(d)} \beta_i^{(d)} (d_{im})^{-\alpha_d} \left\| g_{im}^{(d)} \right\|^2$$
(5.4b)

$$I_{m}^{(d \to d)} = \sum_{\substack{j \in \mathcal{V} \\ j \neq m}} P_{d} A^{(d)} \beta_{j}^{(d)} (d_{jm})^{-\alpha_{d}} \left\| g_{jm}^{(d)} \right\|^{2}$$
(5.4c)

where (5.4a) is the desired signal power, (5.4b) is the interference as a result of V2I links, and (5.4c) is the interference caused by other V2V links. The ergodic spectrum efficiency of the V2I link for the mth vehicle is

$$R_m^{(c)} = \mathbb{E}\left\{\log\left(1 + \gamma_m^{(c)}\right)\right\}$$
(5.5)

also, the V2V link between the kth and mth vehicles is

$$R_m^{(d)} = \mathbb{E}\left\{\log\left(1 + \gamma_m^{(d)}\right)\right\}$$
(5.6)

5.3. Power Allocation

ADAS can benefit from the Mobile edge computing (MEC) concept for accessing their required computational and storage capacities remotely. In such a system, vehicles should be able to upload their sensors' data to MEC servers for further processing. Vehicular communication is the enabler for designing such driver assistance systems, which would operate based on collaborative and remote processing solutions. The requirements for a VANET to support ADAS are V2I links that have large throughputs and V2V links with high reliability.

Employing a BS with multiple antennas will offer a significant opportunity to achieve the desired QoS in V2I links. It is shown in the previous chapter if a BS contains a large array of antennas ($M \rightarrow \infty$) and has the capability of obtaining the perfect CSI through the transmission of training sequences, then by designing $w_m^{(c)}$ as a ZF filter, $\gamma_m^{(c)}$ can increase unboundedly and therefore, any desired capacity can be achieved. However, having a perfect CSI in a vehicular environment that contains high mobility is not practical.

To overcome this limitation, a training sequence needs to be transmitted every few hundred milliseconds for CSI estimation. By extending the results that are achieved in the previous chapter, the V2I spectral efficiency of a system that has a BS with a large array of antennas and ZF filters and also uses a training sequence of length T_c for CSI estimation is equal to

$$R_{m}^{(c)} = \mathbb{E}\left\{ \log\left(1 + \frac{T_{c}P_{c}^{2}\left(A^{(c)}\beta_{m}^{(c)}\right)^{2}(d_{m})^{-2\alpha_{c}}}{\sum_{j\in\mathcal{V}}P_{d}^{2}\left(A^{(c)}\beta_{j}^{(c)}\right)^{2}\left(d_{j}\right)^{-2\alpha_{c}}}\right)\right\} \ge \log\left(1 + \frac{T_{c}P_{c}^{2}}{\mathcal{V}P_{d}^{2}}\right)$$
(5.7)

If the minimum capacity requirement of V2I links is considered as $R_0^{(V2I)}$, then (5.7) can be used to calculate the required transmission power (P_c and P_d) such that $R_m^{(c)} \ge R_0^{(V2I)}$ for all m. On the other hand, the preference for ADAS applications is to monitor a vehicle's surrounding environment for detecting hazardous conditions. Therefore, we consider the V2V links between vehicles that are in each other's LoS. The V2V spectral efficiency given in (5.6) for this system can then be calculated based on $S_{mm}^{(d)} = P_d A^{(d)} (d_{mm})^{-\alpha_d}$ therefore if the minimum capacity requirement of the V2V links is considered as $R_0^{(V2V)}$ then $R_m^{(d)} \ge R_0^{(V2V)}$ for all m.

As $\gamma_m^{(d)}$ is a convex function of $I_m^{(c \to d)}$ and $I_m^{(d \to d)}$ therefore based on Jensen's inequality we have

$$R_{m}^{(d)} = \mathbb{E}\left\{\log\left(1 + \gamma_{m}^{(d)}\right)\right\} \ge \log\left(1 + \frac{S_{mm}^{(d)}}{\mathbb{E}\left\{I_{m}^{(c \to d)} + I_{m}^{(d \to d)}\right\} + N_{0}}\right)$$
(5.8)

For all $i \neq m$, $\{\beta_i^{(d)}\}$, $\{g_{im}^{(d)}\}$ and $\{d_{im}\}$ are independent and $\mathbb{E}\{\beta_i^{(d)}\}=\mathbb{E}\{\|g_{im}^{(d)}\|^2\}=1$. Therefore,

$$R_m^{(d)} \ge \log\left(1 + \frac{P_d(d_{mm})^{-\alpha_d}}{(\mathcal{I}P_c + (\mathcal{V} - 1)P_d)\mathbb{E}\{(d_{im})^{-\alpha_d}\} + \frac{N_0}{A^{(d)}}}\right)$$
(5.9)

Let *L* be the length of the road under the BS coverage, and the spatial positions of the vehicles are modelled as a PPP with intensity λ . Therefore, the probability mass function for the number of vehicles on the road is

$$p\left(\mathcal{I} + \frac{\mathcal{V}}{2} = n\right) = e^{-\lambda L} \frac{(\lambda L)^n}{n!}$$
(5.10)

Moreover, the probability distribution function for the distance between vehicles is

$$f(d) = \left(\frac{1}{\lambda L}\right) e^{-d/\lambda L}$$
(5.11)

which makes

$$\mathbb{E}\{(d_{im})^{-\alpha_d}\} = \int_{d_{mm}}^{L} x^{-\alpha_d} \left(\frac{1}{\lambda L}\right) e^{-\frac{x}{\lambda L}} dx$$
(5.12)

By simultaneously solving (5.7) and (5.9), the minimum required P_d and P_c and the constraint on the distance between vehicles for allowing V2V can be achieved as

$$P_{d} = \frac{\frac{N_{0}}{A^{(d)}}}{\left(2^{R_{0}^{(V2V)}} - 1\right)^{-1} (d_{mm})^{-\alpha_{d}} - \left(\Im\sqrt{\mathcal{V}\left(2^{R_{0}^{(V2I)}} - 1\right)} + \mathcal{V} - 1\right)\int_{d_{mm}}^{L} x^{-\alpha_{d}} \left(\frac{1}{\lambda L}\right) e^{-\frac{x}{\lambda L}} dx}$$
(5.13)
$$P_{c} = \sqrt{\mathcal{V}\left(2^{R_{0}^{(V2I)}} - 1\right)} P_{d}$$
(5.14)

5.4. Summary

Remote computing solutions can enable a highly efficient and cost-effective ADAS for future generation vehicles, and vehicular communication is the enabler for such a method through allowing vehicles to communicate with remote servers by uploading their sensor data for further processing. Large capacity links for V2I connections and high-reliability links for V2V connections are the QoS requirements for a VANET to support ADAS. The V2V and V2I links can respectively be supported by D2D and Device-to-eNodeB links in a cellular-based V2X. However, resource management is very crucial, as the cellular communication and the in-band D2D links cause cross interferences on one another, as they both transmit over the same frequency spectrum. In this paper, we used the concept of massive MIMO to cancel the interference in the V2I links. The underlay D2D communication on the uplink spectrum is employed for V2V links. We derived the equations from calculating the minimum transmission power to achieve the required spectral efficiency for both V2I and V2V links.

Chapter 6 : RESULTS AND DISCUSSION

6.1. Introduction

In this chapter, we provide a numerical analysis to demonstrate the effect of transmission power on the V2I and V2V spectral efficiencies. The specific parameters used are summarized in Table 6.1. The parameters are according to 3GPP TR 36.885 [86].

BS Coverage L	500 m
The density of vehicles λ	1/50 m ⁻¹
V2I pathloss reference $A^{(c)}$	15.3 dB
V2V pathloss reference $A^{(d)}$	38.5 dB
V2I pathloss pathloss exponent α_c	3.76
V2V pathloss pathloss exponent α_d	4.37
V2I Lognormal Shadowing σ_{sc}	7 dB
V2V Lognormal Shadowing σ_{sd}	3 dB
Noise PSD	-114 dBm/Hz
Channel bandwidth	10 MHz

Table 6.1. Simulation parameters

6.2. Simulation Results and Discussions

Vehicles are distributed according to homogeneous PPP with intensity λ . The shadowing is lognormal with a deviation σ_{sc} (dB) and σ_{sd} (dB) for V2I and V2V accordingly. The difference is a result of antenna height in the BS and vehicle receiver. The BS has a large array of antennas, and a CSI is estimated based on training sequences. Based on the given values

$$\mathbb{E}\{(d_{im})^{-\alpha_d}\} = \int_{d_{mm}}^{500} 0.1 x^{-4.37} e^{-0.1x} \, dx \tag{6.1}$$

From (5.7) we can write

$$P_c = \sqrt{\mathcal{V}\left(2^{R_0^{(\text{V2I})}} - 1\right)} P_d \tag{6.2}$$

This shows for maintaining QoS on V2I links, as the number of V2V links increases, the transmission power should also be increased. Also (6.2) provides a guideline for the power ratio between different transmission links. Figure 6.1 shows the power ratio of V2I to V2V links for different spectral efficiencies. As it is seen, more transmission power is required at a higher $R_0^{(V2I)}$ for maintaining the QoS affected by increasing the number of V2V links.



Figure 6.1. The transmission power ratio of V2I to V2V links for different spectral efficiencies values of the V2I link.

Figure 6.2 shows the spectral efficiencies of V2V links for the power ratio of 10. In this figure, $R_m^{(d)}$ has been calculated using (5.12) and (6.1) for the $P_c = 23 \ dBm$ and $P_d = 13 \ dBm$. As it can be seen by increasing the number of V2V communications, the QoS in the V2V links becomes degraded. This is as a result of interferences, which are added to the system from each extra V2V communication between vehicles.



Figure 6.2. The relation between V2V links spectral efficiencies and the ratio of V2V to V2I links.

Chapter 7 : CONCLUSION

In this dissertation, a method to calculate the minimum transmission power for providing the required spectral efficiency and mitigating the interference between the Cellular User Equipments and Device-to-Device User Equipments of a cellular network was proposed. We only consider D2D links between devices that are in each other's line-of-sight. Other than that, no other limitations have been considered. Furthermore, we expand this idea to a cellular-based vehicular communication system, which employs the concept of D2D communication for establishing V2V transmission links.

In chapter 1, the motivation for this research and an overview of the dissertation was given. Vehicular communications and remote processing solutions can be considered as enablers for designing cost-effective ADAS in future vehicles. D2D communication is a suitable candidate for being used in a cellular-based vehicular communication system to support V2V links. D2D communication links, generally, should have minimal or no interference effects on the nearby ongoing cellular communication, and therefore, resource management is a requirement for achieving this goal. Nearby mobile devices can communicate directly without their information traversing through the cellular base station or the evolved NodeB, and as a result, D2D links can provide low-latency, high data-rate communications while consuming relatively less amount of power.

The existing literature on D2D communications and the cellular-based vehicular communication systems was reviewed in chapter 2. D2D communications are considered as one of the solutions for improving spectrum utilization in the next generation of cellular networks. D2D communications can increase the overall capacity of a cellular network. They can also enhance the coverage of a cellular network, especially on the edges of a cell, by allowing BS to access the out-of-coverage DUEs through CUEs that receive adequate levels of signal power. As the distances between DUE's are shorter, D2D communication links can use less power to transmit higher data-rates with shorter delays. However, by adding D2D users to a cellular network, radio resource management becomes a critical part of achieving the required level of QoS. Many applications can be considered for D2D communications, such as high-

volume content sharing of videos, pictures, and gaming. D2D communications are also well suited for establishing V2V links between nearby vehicles, and it has been shown in previous studies that D2D-assisted vehicular communication can surpass traditional V2V techniques.

In chapter 3, the mathematical background required for modelling a D2D communication system, including the effects of multipath propagation shortcomings, was discussed. Based on the variation of the received signal strength, impairments can be categorized into two groups of large and small-scale impairments. Large-scale effects happen over large distances or during long time intervals (compared with the carrier frequency), and small-scale effects take place over small distances or short time intervals. Impairments such as path loss, noise, shadowing, and fading were covered. Moreover, the use of multiple antenna systems for overcoming fading degradation and combating interference effects was discussed.

An underlay of D2D communications that share the UL spectrum of a cellular network is investigated in chapter 4. A careful power control mechanism is a requirement for underlay D2D communication systems to control the level of interference, and thus to obtain the desired spectral efficiencies. We derived formulas to calculate the minimum amount of transmission power for both CUEs and DUEs to achieve such goals. We further use the concept of massive MIMO to minimise the interference effects of the user-to-cellular links.

A cellular vehicular communication system is discussed in chapter 5 that uses the concept of D2D communication for establishing V2V links. Therefore, nearby vehicles would be able to communicate with one another without their data traversing the core network. Furthermore, the V2I links were modelled by Device-to-eNodeB communication links. The formulas that were derived in the previous chapter have been expanded for this model, which resulted in equations to calculate the minimum transmission power for achieving the required spectral efficiencies of both V2I and V2V links.

To demonstrate the effect of transmission power on the spectral efficiencies of V2I and V2V links, we perform numerical analysis in chapter 6. Simulations are done based on 3GPP TR 36.885 parameters. It was shown that to maintain the required level of QoS, the transmission power should be increased as the number of V2V links increases. Furthermore, obtaining a higher $R_0^{(V2I)}$ requires spending more power for transmission. We also showed that the QoS of

V2V links degrades by increasing the number of V2V links. This is as a result of the additional interferences, which are introduced to the system by each extra V2V communication link.

In summary, cellular-based vehicular communication can pave the way for designing highly efficient and cost-effective driver assistance systems based on collaborative and remote processing solutions. The main transmission links for vehicular communication systems are V2V and V2I. D2D communication links are considered for establishing V2V links, and cellular communication links are employed for V2I links. An underlay in-band D2D communication is considered, where DUE's reuse the cellular UL spectrum. Statistical methods are employed to find boundaries on the allocated transmission powers for achieving sufficient spectral efficiencies in V2I and V2V links without considering any prior knowledge on vehicles' locations or the channel state information. The capacity of such a system is directly proportional to the energy of a transmitted signal, and in fact, achieving the desired level of QoS requires careful control of transmission power for all the radio sources within a system.

In this research, we only consider devices with a single antenna. Therefore, as future work, systems with multiple-antennas devices can be considered. The other assumption that has been made with regards to employing massive MIMO was considering favourable propagation. However, this may not be held when we have a channel with correlated scatters. This topic can also be considered in future research.

REFERENCES

- [1] G. Velez and O. Otaegui, "Embedding vision-based advanced driver assistance systems: a survey," *IET Intelligent Transport Systems*, vol. 11, no. 3, pp. 103-112, 2017, doi: 10.1049/iet-its.2016.0026.
- [2] A. Keivani, F. Ghayoor, and J.-R. Tapamo, "Collaborative Mobile Edge Computing in eV2X: A Solution for Low-Cost Driver Assistance Systems," *Wireless Personal Communications*, 04/24 2019, doi: 10.1007/s11277-019-06401-2.
- [3] P. Pirinen, "A brief overview of 5G research activities," in *1st International Conference* on 5G for Ubiquitous Connectivity, 26-28 Nov. 2014 2014, pp. 17-22, doi: 10.4108/icst.5gu.2014.258061.
- [4] G. Liu and D. Jiang, "5G: Vision and Requirements for Mobile Communication System towards Year 2020," *Chinese Journal of Engineering*, vol. 2016, pp. 1-8, 04/06 2016, doi: 10.1155/2016/5974586.
- [5] A. Pattavina, S. Quadri, and V. Trecordi, "Spectral efficiency improvement of enhanced assignment techniques in cellular networks," in *Proceedings of GLOBECOM'96. 1996 IEEE Global Telecommunications Conference*, 18-28 Nov. 1996 1996, vol. 2, pp. 1031-1035 vol.2, doi: 10.1109/GLOCOM.1996.587588.
- [6] A. Abrol and R. K. Jha, "Power Optimization in 5G Networks: A Step Towards GrEEn Communication," *IEEE Access*, vol. 4, pp. 1355-1374, 2016, doi: 10.1109/ACCESS.2016.2549641.
- [7] F. Jameel, Z. Hamid, F. Jabeen, S. Zeadally, and M. A. Javed, "A Survey of Device-to-Device Communications: Research Issues and Challenges," *IEEE Communications Surveys & Tutorials*, vol. 20, no. 3, pp. 2133-2168, 2018, doi: 10.1109/COMST.2018.2828120.
- [8] P. Mach, Z. Becvar, and T. Vanek, "In-Band Device-to-Device Communication in OFDMA Cellular Networks: A Survey and Challenges," *IEEE Communications Surveys & amp Tutorials*, vol. 17, p. 38, 11/01 2015, doi: 10.1109/COMST.2015.2447036.
- [9] A. T. Gamage, H. Liang, R. Zhang, and X. Shen, "Device-to-device communication underlaying converged heterogeneous networks," *IEEE Wireless Communications*, vol. 21, no. 6, pp. 98-107, 2014, doi: 10.1109/MWC.2014.7000977.
- [10] C. Gao *et al.*, "Evaluating the Impact of User Behavior on D2D Communications in Millimeter-Wave Small Cells," *IEEE Transactions on Vehicular Technology*, vol. 66, no. 7, pp. 6362-6377, 2017, doi: 10.1109/TVT.2016.2642127.
- [11] N. Anjum, Z. Yang, H. Saki, M. Kiran, and M. Shikh-Bahaei, "Device-to-Device (D2D) Communication as a Bootstrapping System in a Wireless Cellular Network," *IEEE Access*, vol. PP, pp. 1-1, 01/07 2019, doi: 10.1109/ACCESS.2019.2890987.
- [12] K. Ali, H. Nguyen, Q.-T. Vien, P. Shah, and Z. Chu, "Disaster Management Using D2D Communication With Power Transfer and Clustering Techniques," *IEEE Access*, vol. PP, pp. 1-1, 01/15 2018, doi: 10.1109/ACCESS.2018.2793532.
- [13] J. Gui and K. Zhou, "Cellular throughput optimization by game-based power adjustment and outband D2D communication," *EURASIP Journal on Wireless Communications and Networking*, vol. 2018, no. 1, p. 254, 2018/10/25 2018, doi: 10.1186/s13638-018-1275-2.
- [14] W. Sun, E. G. Ström, F. Brännström, Y. Sui, and K. C. Sou, "D2D-based V2V communications with latency and reliability constraints," in 2014 IEEE Globecom

Workshops (GC Wkshps), 8-12 Dec. 2014 2014, pp. 1414-1419, doi: 10.1109/GLOCOMW.2014.7063632.

- [15] W. He, G. Yan, and L. D. Xu, "Developing Vehicular Data Cloud Services in the IoT Environment," *Industrial Informatics, IEEE Transactions on*, vol. 10, pp. 1587-1595, 05/01 2014, doi: 10.1109/TII.2014.2299233.
- [16] H. Albasry and Q. Z. Ahmed, "Network-Assisted D2D Discovery Method by Using Efficient Power Control Strategy," in 2016 IEEE 83rd Vehicular Technology Conference (VTC Spring), 15-18 May 2016 2016, pp. 1-5, doi: 10.1109/VTCSpring.2016.7504365.
- [17] M. Naslcheraghi, L. Marandi, and S. A. Ghorashi, "A novel device-to-device discovery scheme for underlay cellular networks," in 2017 Iranian Conference on Electrical Engineering (ICEE), 2-4 May 2017 2017, pp. 2106-2110, doi: 10.1109/IranianCEE.2017.7985409.
- [18] S. Jung and S. Chang, "A discovery scheme for device-to-device communications in synchronous distributed networks," in *16th International Conference on Advanced Communication Technology*, 16-19 Feb. 2014 2014, pp. 815-819, doi: 10.1109/ICACT.2014.6779073.
- [19] L. Wei, R. Hu, Y. Qian, and G. Wu, "Enable Device-to-Device Communications Underlaying Cellular Networks: Challenges and Research Aspects," *Communications Magazine, IEEE*, vol. 52, pp. 90-96, 06/01 2014, doi: 10.1109/MCOM.2014.6829950.
- [20] C. Vlachos, H. Elshaer, J. Chen, V. Friderikos, and M. Dohler, "Bio-Inspired Resource Allocation for Relay-Aided Device-to-Device Communications," 06/15 2016.
- [21] L. Ying-Dar and H. Yu-Ching, "Multihop cellular: a new architecture for wireless communications," in *Proceedings IEEE INFOCOM 2000. Conference on Computer Communications. Nineteenth Annual Joint Conference of the IEEE Computer and Communications Societies (Cat. No.00CH37064)*, 26-30 March 2000 2000, vol. 3, pp. 1273-1282 vol.3, doi: 10.1109/INFCOM.2000.832516.
- [22] P. Janis, "Device-to-Device Communication Underlaying Cellular Communications Systems," *Int'l J. of Communications, Network and System Sciences,* vol. 02, pp. 169-178, 01/01 2009, doi: 10.4236/ijcns.2009.23019.
- [23] "IEEE Standard for Information technology—Telecommunications and information exchange between systems Local and metropolitan area networks—Specific requirements - Part 11: Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications," *IEEE Std 802.11-2016 (Revision of IEEE Std 802.11-2012)*, pp. 1-3534, 2016, doi: 10.1109/IEEESTD.2016.7786995.
- [24] G. A. Safdar, M. Ur-Rehman, M. Muhammad, M. A. Imran, and R. Tafazolli, "Interference Mitigation in D2D Communication Underlaying LTE-A Network," *IEEE Access*, vol. 4, pp. 7967-7987, 2016, doi: 10.1109/ACCESS.2016.2621115.
- [25] Z. Liu, T. Peng, S. Xiang, and W. Wang, "Mode selection for device-to-device (D2D) communication under LTE-advanced networks," in *2012 IEEE International Conference on Communications (ICC)*, 2012: IEEE, pp. 5563-5567.
- [26] C. Vitale, V. Mancuso, and G. Rizzo, "Modelling D2D communications in cellular access networks via Coupled Processors," in 2015 7th International Conference on Communication Systems and Networks (COMSNETS), 6-10 Jan. 2015 2015, pp. 1-8, doi: 10.1109/COMSNETS.2015.7098687.
- [27] X. Lin, J. G. Andrews, and A. Ghosh, "Spectrum sharing for device-to-device communication in cellular networks," *IEEE Transactions on Wireless Communications*, vol. 13, no. 12, pp. 6727-6740, 2014.

- [28] S. Sobhi-Givi, A. Khazali, H. Kalbkhani, M. G. Shayesteh, and V. Solouk, "Resource allocation and power control for underlay device-to-device communication in fractional frequency reuse cellular networks," *Telecommunication Systems*, vol. 65, 12/27 2016, doi: 10.1007/s11235-016-0256-1.
- [29] C.-H. Yu, K. Doppler, C. B. Ribeiro, and O. Tirkkonen, "Resource Sharing Optimization for Device-to-Device Communication Underlaying Cellular Networks," *Wireless Communications, IEEE Transactions on*, vol. 10, pp. 2752-2763, 09/01 2011, doi: 10.1109/TWC.2011.060811.102120.
- [30] A. Orsino *et al.*, "Characterization of User Mobility in Cellular-Assisted D2D Systems," *IEEE Vehicular Technology Magazine*, vol. 11, 04/01 2016, doi: 10.1109/MVT.2016.2550002.
- [31] X. Xu, Y. Zhang, Z. Sun, Y. Hong, and X. Tao, "Analytical Modeling of Mode Selection for Moving D2D-Enabled Cellular Networks," *IEEE Communications Letters*, vol. 20, no. 6, pp. 1203-1206, 2016, doi: 10.1109/LCOMM.2016.2552171.
- [32] M. Huss, M. Waqas, A. Y. Ding, Y. Li, S. Tarkoma, and J. ott, "Security and Privacy in Device-to-Device (D2D) Communication: A Review," *IEEE Communications Surveys* & *Tutorials*, vol. PP, 01/09 2017, doi: 10.1109/COMST.2017.2649687.
- [33] S. Sekander, H. Tabassum, and E. Hossain, "Multi-Tier Drone Architecture for 5G/B5G Cellular Networks: Challenges, Trends, and Prospects," *IEEE Communications Magazine*, vol. 56, 11/19 2017, doi: 10.1109/MCOM.2018.1700666.
- [34] Y. Xu, F. Liu, and P. Wu, "Interference management for D2D communications in heterogeneous cellular networks," *Pervasive and Mobile Computing*, vol. 51, 10/01 2018, doi: 10.1016/j.pmcj.2018.10.005.
- [35] K. Doppler, M. Rinne, P. Janis, C. Ribeiro, and K. Hugl, *Device-to-Device Communications; Functional Prospects for LTE-Advanced Networks*. 2009, pp. 1-6.
- [36] G. Fodor *et al.*, "Design aspects of network assisted device-to-device communications," *IEEE Communications Magazine*, vol. 50, no. 3, pp. 170-177, 2012, doi: 10.1109/MCOM.2012.6163598.
- [37] J. Liu, N. Kato, J. Ma, and N. Kadowaki, "Device-to-device communication in LTEadvanced networks: A survey," *IEEE Communications Surveys & Tutorials*, vol. 17, no. 4, pp. 1923-1940, 2014.
- [38] P. Phunchongharn, E. Hossain, and D. I. Kim, "Resource allocation for device-to-device communications underlaying LTE-advanced networks," *IEEE Wireless Communications*, vol. 20, no. 4, pp. 91-100, 2013, doi: 10.1109/MWC.2013.6590055.
- [39] Y. Jiang, Q. Liu, F. Zheng, X. Gao, and X. You, *Energy Efficient Joint Resource Allocation and Power Control for D2D Communications*. 2016, pp. 6119-6127.
- [40] C. Yang, J. Li, P. Semasinghe, E. Hossain, S. M. Perlaza, and Z. Han, "Distributed Interference and Energy-Aware Power Control for Ultra-Dense D2D Networks: A Mean Field Game," *IEEE Transactions on Wireless Communications*, vol. 16, no. 2, pp. 1205-1217, 2017, doi: 10.1109/TWC.2016.2641959.
- [41] N. Lee, X. Lin, J. G. Andrews, and R. W. Heath, "Power control for D2D underlaid cellular networks: Modeling, algorithms, and analysis," *IEEE Journal on Selected Areas in Communications*, vol. 33, no. 1, pp. 1-13, 2014.
- [42] G. Yu, L. Xu, D. Feng, R. Yin, G. Y. Li, and Y. Jiang, "Joint mode selection and resource allocation for device-to-device communications," *IEEE transactions on communications*, vol. 62, no. 11, pp. 3814-3824, 2014.
- [43] D. Feng, L. Lu, Y. Yuan-Wu, G. Y. Li, G. Feng, and S. Li, "Device-to-Device Communications Underlaying Cellular Networks," *IEEE Transactions on*

Communications, vol. 61, no. 8, pp. 3541-3551, 2013, doi: 10.1109/TCOMM.2013.071013.120787.

- [44] B. Kaufman, J. Lilleberg, and B. Aazhang, "Spectrum sharing scheme between cellular users and ad-hoc device-to-device users," *IEEE Transactions on Wireless Communications*, vol. 12, no. 3, pp. 1038-1049, 2013.
- [45] M. R. Mili, P. Tehrani, and M. Bennis, "Energy-efficient power allocation in OFDMA D2D communication by multiobjective optimization," *IEEE Wireless Communications Letters*, vol. 5, no. 6, pp. 668-671, 2016.
- [46] R. AliHemmati, B. Liang, M. Dong, G. Boudreau, and S. H. Seyedmehdi, "Power Allocation for Underlay Device-to-Device Communication Over Multiple Channels," *IEEE Transactions on Signal and Information Processing over Networks*, vol. 4, no. 3, pp. 467-480, 2018, doi: 10.1109/TSIPN.2017.2742941.
- [47] R. Li, P. Hong, K. Xue, M. Zhang, and T. Yang, "Energy-Efficient Resource Allocation for High-Rate Underlay D2D Communications with Statistical CSI: A One-to-Many Strategy," *IEEE Transactions on Vehicular Technology*, vol. 69, no. 4, pp. 4006-4018, 2020.
- [48] S. Liu, Y. Wu, L. Li, X. Liu, and W. Xu, "A Two-Stage Energy-Efficient Approach for Joint Power Control and Channel Allocation in D2D Communication," *IEEE Access*, vol. 7, pp. 16940-16951, 2019, doi: 10.1109/ACCESS.2019.2894003.
- [49] A. Abdallah, M. M. Mansour, and A. Chehab, "Power Control and Channel Allocation for D2D Underlaid Cellular Networks," *IEEE Transactions on Communications*, vol. 66, no. 7, pp. 3217-3234, 2018, doi: 10.1109/TCOMM.2018.2812731.
- [50] J. Huang, C. Xing, and M. Guizani, "Power Allocation for D2D Communications With SWIPT," *IEEE Transactions on Wireless Communications*, vol. 19, no. 4, pp. 2308-2320, 2020, doi: 10.1109/TWC.2019.2963833.
- [51] I. lu, G. Li, A. Swindlehurst, A. Ashikhmin, and R. Zhang, *An Overview of Massive MIMO: Benefits and Challenges*. 2014, pp. 742-758.
- [52] T. L. Marzetta, "Massive MIMO: An Introduction," *Bell Labs Technical Journal*, vol. 20, pp. 11-22, 2015, doi: 10.15325/BLTJ.2015.2407793.
- [53] S. Yang and L. Hanzo, "Fifty Years of MIMO Detection: The Road to Large-Scale MIMOs," *IEEE Communications Surveys & Tutorials*, vol. 17, no. 4, pp. 1941-1988, 2015, doi: 10.1109/COMST.2015.2475242.
- [54] X. Lin, R. Heath, and J. Andrews, *The Interplay between Massive MIMO and Underlaid D2D Networking*. 2014.
- [55] A. He, L. Wang, Y. Chen, K.-K. Wong, and M. Elkashlan, "Spectral and energy efficiency of uplink D2D underlaid massive MIMO cellular networks," *IEEE Transactions on Communications*, vol. 65, no. 9, pp. 3780-3793, 2017.
- [56] H. Xu, W. Xu, Z. Yang, J. Shi, and M. Chen, "Pilot reuse among D2D users in D2D underlaid massive MIMO systems," *IEEE Transactions on Vehicular Technology*, vol. 67, no. 1, pp. 467-482, 2017.
- [57] A. Ghazanfari, E. Björnson, and E. G. Larsson, "Optimized power control for massive MIMO with underlaid D2D communications," *IEEE Transactions on Communications*, vol. 67, no. 4, pp. 2763-2778, 2018.
- [58] S. Guo and X. Zhou, "Robust Resource Allocation With Imperfect Channel Estimation in NOMA-Based Heterogeneous Vehicular Networks," *IEEE Transactions on Communications*, vol. 67, no. 3, pp. 2321-2332, 2019, doi: 10.1109/TCOMM.2018.2885999.

- [59] S. Yousefi, S. Bastani, and M. Fathy, On the Performance of Safety Message Dissemination in Vehicular Ad Hoc Networks. 2007, pp. 377-390.
- [60] M. Gholibeigi, N. Sarrionandia, M. Karimzadeh, M. Baratchi, H. v. d. Berg, and G. Heijenk, "Reliable vehicular broadcast using 5G device-to-device communication," in 2017 10th IFIP Wireless and Mobile Networking Conference (WMNC), 25-27 Sept. 2017 2017, pp. 1-8, doi: 10.1109/WMNC.2017.8248846.
- [61] A. Bazzi, B. M. Masini, A. Zanella, and I. Thibault, "On the Performance of IEEE 802.11p and LTE-V2V for the Cooperative Awareness of Connected Vehicles," *IEEE Transactions on Vehicular Technology*, vol. 66, no. 11, pp. 10419-10432, 2017, doi: 10.1109/TVT.2017.2750803.
- [62] H. Seo, K. Lee, S. Yasukawa, Y. Peng, and P. Sartori, "LTE evolution for vehicle-toeverything services," *IEEE Communications Magazine*, vol. 54, no. 6, pp. 22-28, 2016, doi: 10.1109/MCOM.2016.7497762.
- [63] A. Keivani, F. Ghayoor, and J. Tapamo, "A vision-based driver assistance system using collaborative edge computing," in 2017 Global Wireless Summit (GWS), 15-18 Oct. 2017 2017, pp. 160-164, doi: 10.1109/GWS.2017.8300480.
- [64] G. Araniti, C. Campolo, M. Condoluci, A. Iera, and A. Molinaro, "LTE for vehicular networking: a survey," *IEEE communications magazine*, vol. 51, no. 5, pp. 148-157, 2013.
- [65] A. Asadi, Q. Wang, and V. Mancuso, "A survey on device-to-device communication in cellular networks," *IEEE Communications Surveys & Tutorials*, vol. 16, no. 4, pp. 1801-1819, 2014.
- [66] W. Sun, E. G. Ström, F. Brännström, K. C. Sou, and Y. Sui, "Radio resource management for D2D-based V2V communication," *IEEE Transactions on Vehicular Technology*, vol. 65, no. 8, pp. 6636-6650, 2015.
- [67] X. Cao, L. Liu, Y. Cheng, L. X. Cai, and C. Sun, "On optimal device-to-device resource allocation for minimizing end-to-end delay in VANETs," *IEEE transactions on vehicular technology*, vol. 65, no. 10, pp. 7905-7916, 2016.
- [68] X. Cheng, L. Yang, and X. Shen, "D2D for Intelligent Transportation Systems: A Feasibility Study," *IEEE Transactions on Intelligent Transportation Systems*, vol. 16, no. 4, pp. 1784-1793, 2015, doi: 10.1109/TITS.2014.2377074.
- [69] N. Cheng *et al.*, "Performance analysis of vehicular device-to-device underlay communication," *IEEE Transactions on vehicular technology*, vol. 66, no. 6, pp. 5409-5421, 2016.
- [70] L. Liang, G. Y. Li, and W. Xu, "Resource Allocation for D2D-Enabled Vehicular Communications," *IEEE Transactions on Communications*, vol. 65, no. 7, pp. 3186-3197, 2017, doi: 10.1109/TCOMM.2017.2699194.
- [71] W. Sun, D. Yuan, E. G. Ström, and F. Brännström, "Cluster-based radio resource management for D2D-supported safety-critical V2X communications," *IEEE Transactions on Wireless Communications*, vol. 15, no. 4, pp. 2756-2769, 2015.
- [72] A. Osseiran *et al.*, "Scenarios for 5G mobile and wireless communications: the vision of the METIS project," *IEEE communications magazine*, vol. 52, no. 5, pp. 26-35, 2014.
- [73] K. K. Nguyen, T. Q. Duong, N. A. Vien, N. Le-Khac, and L. D. Nguyen, "Distributed Deep Deterministic Policy Gradient for Power Allocation Control in D2D-Based V2V Communications," *IEEE Access*, vol. 7, pp. 164533-164543, 2019, doi: 10.1109/ACCESS.2019.2952411.

- [74] C. Guo, L. Liang, and G. Y. Li, "Resource allocation for high-reliability low-latency vehicular communications with packet retransmission," *IEEE Transactions on Vehicular Technology*, vol. 68, no. 7, pp. 6219-6230, 2019.
- [75] C. Guo, L. Liang, and G. Y. Li, "Resource allocation for vehicular communications with low latency and high reliability," *IEEE Transactions on Wireless Communications*, vol. 18, no. 8, pp. 3887-3902, 2019.
- [76] X. Peng, H. Zhou, B. Qian, K. Yu, F. Lyu, and W. Xu, "Enabling Security-Aware D2D Spectrum Resource Sharing for Connected Autonomous Vehicles," *IEEE Internet of Things Journal*, vol. 7, no. 5, pp. 3799-3811, 2020.
- [77] C. He, Q. Chen, C. Pan, X. Li, and F.-C. Zheng, "Resource allocation schemes based on coalition games for vehicular communications," *IEEE Communications Letters*, vol. 23, no. 12, pp. 2340-2343, 2019.
- [78] H. Xiao, D. Zhu, and A. T. Chronopoulos, "Power allocation with energy efficiency optimization in cellular D2D-based V2X communication network," *IEEE Transactions* on Intelligent Transportation Systems, vol. 21, no. 12, pp. 4947-4957, 2019.
- [79] Z. Liu, X. Han, Y. Liu, and Y. Wang, "D2D-based vehicular communication with delayed CSI feedback," *IEEE Access*, vol. 6, pp. 52857-52866, 2018.
- [80] M. Belleschi, G. Fodor, and A. Abrardo, "Performance analysis of a distributed resource allocation scheme for D2D communications," in 2011 ieee globecom workshops (gc wkshps), 2011: IEEE, pp. 358-362.
- [81] C.-H. Yu, K. Doppler, C. B. Ribeiro, and O. Tirkkonen, "Resource sharing optimization for device-to-device communication underlaying cellular networks," *IEEE Transactions on Wireless communications*, vol. 10, no. 8, pp. 2752-2763, 2011.
- [82] A. Goldsmith, *Wireless communications*. Cambridge university press, 2005.
- [83] J. G. Proakis and M. Salehi, *Digital communications*. McGraw-Hill., 2008.
- [84] H. Q. Ngo, E. G. Larsson, and T. L. Marzetta, "Energy and spectral efficiency of very large multiuser MIMO systems," *IEEE Transactions on Communications*, vol. 61, no. 4, pp. 1436-1449, 2013.
- [85] R. García-Pelayo, "Distribution of distance in the spheroid," *Journal of Physics A: Mathematical and General*, vol. 38, no. 16, p. 3475, 2005.
- [86] Scenarios, "Requirements KPIs for 5G Mobile and Wireless System, Standard ICT-317669-METIS/D1.1," Apr 2013.